SPARK NEW ZEALAND LIMITED Spark NZ

REVERA LIMITED SPARK FINANCE LIMITED SPARK NEW ZEALAND TRADING LIMITED TELECOM ENTERPRISES LIMITED TELECOM PACIFIC LIMITED TCNZ (UNITED KINGDOM) SECURITIES LIMITED Guaranteeing Subsidiaries

THE NEW ZEALAND GUARDIAN TRUST COMPANY LIMITED Supervisor

DEED AMENDING AND RESTATING TRUST DEED DATED 25 OCTOBER 1988

RUSSELL MºVEAGH

A November DEED dated

2015

PARTIES

SPARK NEW ZEALAND LIMITED ("Spark NZ")

REVERA LIMITED SPARK FINANCE LIMITED SPARK NEW ZEALAND TRADING LIMITED TELECOM ENTERPRISES LIMITED TELECOM PACIFIC LIMITED TCNZ (UNITED KINGDOM) SECURITIES LIMITED ("Guaranteeing Subsidiaries")

THE NEW ZEALAND GUARDIAN TRUST COMPANY LIMITED ("Supervisor")

INTRODUCTION

- A. Spark NZ and the Supervisor have executed a trust deed relating to the constitution and issue of unsecured stock dated 25 October 1988, as amended and restated on 19 September 2012 and 18 September 2014 ("**Original Deed**").
- B. Spark NZ and the Supervisor wish to amend and restate the Original Deed on the terms set out in this deed.
- C. For the purposes of clause 24.1 of the Original Deed the Supervisor and Spark NZ are of the opinion that the amendments contained in this deed are:
 - (i) of a formal or technical nature; or
 - (ii) made to comply with the requirements of the Financial Markets Conduct Act 2013.

COVENANTS

1. INTERPRETATION

1.1 **Definitions**:

- (a) Words and expressions that are defined in the Original Deed have the same meanings when used in this deed, unless the context otherwise requires.
- (b) In this deed, "Effective Date" means the date Spark Finance Limited, as issuer under the Original Deed, elects in accordance with clause 19(1)(a) of schedule 4 of the FMC Act and notifies to the Supervisor.

1.2 Interpretation:

- (a) Headings are inserted for convenience only and do not affect the interpretation of this deed.
- (b) Unless the context otherwise requires the singular includes the plural and vice versa and words denoting individuals include other persons and vice versa.
- (c) A reference to any document includes reference to that document as modified, novated, supplemented, varied or replaced from time to time.

(d) A reference to any party to a document includes its successors and permitted assigns.

2. AMENDMENT AND RESTATEMENT OF THE ORIGINAL DEED

- 2.1 **Amendment and restatement**: With effect from the Effective Date, the Original Deed shall be amended and restated in the form of the document set out in the schedule.
- 2.2 **Confirmation**: Except to the extent amended and restated by this deed, the Original Deed continues in full force and effect.
- 2.3 **Guaranteeing Subsidiaries**: Each Guaranteeing Subsidiary:
 - (a) agrees to and approves of the amendments to the Original Deed contained in this deed; and
 - (b) confirms and agrees that its liabilities and obligations as a Guaranteeing Subsidiary under the Original Deed continue in full force and effect.

3. COUNTERPARTS

3.1 This deed may be signed in counterpart copies, both of which will together constitute one and the same instrument, and either of the parties may execute this deed by signing any such counterpart.

4. GOVERNING LAW

4.1 This deed shall be governed by, and construed in accordance with, the laws of New Zealand.

EXECUTION

Executed as a deed

Spark NZ

SPARK NEW ZEALAND LIMITED by:

Signature of authorised signatory

Jolie Hodson Name of authorised signatory

Signature of authorised Name of authorised signato

and witnessed by: Signature of witness Name of with Occupat

City/town of residence

Supervisor

THE NEW ZEALAND GUARDIAN TRUST COMPANY LIMITED by:

Signature of authorised signatory

Signature of authorised signatory

Name of authorised signatory

Name of authorised signatory

and witnessed by:

Signature of witness

Name of witness

Occupation

City/town of residence

EXECUTION

Executed as a deed

Spark NZ

SPARK NEW ZEALAND LIMITED by:

Signature of authorised signatory

Signature of authorised signatory

Name of authorised signatory

Name of authorised signatory

and witnessed by:

Signature of witness

Name of witness

Occupation

City/town of residence

Supervisor

THE ZEALAND GUARDIAN NEV TRUST COMPANY LIMITED by: Signature of authorised signatory ILMA JR Ö

Name of authorised signatory

and witnessed by: Signature of witness

Joanne Mulholland Name oOlierst Assistant Perpetual Guardian Occupation

City/town of residence

Signature of authorised signatory

Lakes. 114 10

Name of authorised signatory

Guaranteeing Subsidiaries

REVERA LIMITED by:

pd Signature of director/authorised signatory

Mie Hopke

Name of director/authorised signatory

and witnessed by: Signature of witness 1 Name of witness Solicitor Occupati and

Signature of director/authorised signatory Name of director/authorised signatory

Sarah Gold Solicitor Auckland

City/town of residence

SPARK FINANCE LIMITED by:

Signature of director/authorised signatory Toke

Name of director/authorised signatory

and witnessed by: Signature of witness Sarah Name of witne Occupat

Signature of director/autholised signatory States Kyth. Name of director/authorised signatory

> Sarah Gold Solicitor Auckland

City/town of residence

SPARK NEW ZEALAND TRADING LIMITED by:

odhe Signature of director/authorised signatory Name of director/authorised signatory

and witnessed by: Signature of witness Sarah Gold Name of witness Occupation cklan

City/town of residence

TELECOM ENTERPRISES LIMITED by:

och

Signature of director/authorised signatory

VOIL TT 0 Name of director/authorised signatory

and witnessed by: 20 Signature of witness Gold arah Nameof Allicitor Occupat ckland

City/town of residence

Signature of director/authorised signatory S 0 QV Name of director/authorised signator

Signature of director/authorised signatory C Name of director/authorised signator

TELECOM PACIFIC LIMITED by:

Signature of director/authorised signatory

Jolie Hedson

Name of director/authorised signatory

and witnessed by: Signature of witness

Name of witr Inch Occupat

City/town of residence

TCNZ (UNITED KINGDOM) SECURITIES LIMITED by:

Signature of director/authorised signatory

Signature of director/authorised signatory

Name of director/authorised signatory

Name of director/authorised signatory

and witnessed by:

Signature of witness

Name of witness

Occupation

City/town of residence

Signature of director/authorised signatory

TELECOM PACIFIC LIMITED by:

Signature of director/authorised signatory

Signature of director/authorised signatory

Name of director/authorised signatory

Name of director/authorised signatory

and witnessed by:

Signature of witness

Name of witness

Occupation

 \bigcirc

City/town of residence

TCNZ (UNITED KINGDOM) SECURITIES

finite formation and the second s

Signature of director/authorised signatory

JASON READER Name of director/authorised signatory

Signature of director/authorised signatory

MARTIN PALMER Name of director/authorised signatory

and witnessed by M

Signature of witness

LEE Moore Name of witness

Manager

Brastsl, UK City/town of residence

SCHEDULE

Amended and Restated Trust Deed

Trust Deed Relating to the Constitution and Issue of Unsecured Stock

Spark New Zealand Limited (Spark NZ)

The New Zealand Guardian Trust Company Limited (Supervisor)

CONTENTS

PART I		1	
INTERPRETATION AND APPOINTMENT OF SUPERVISOR			
1	DEFINITIONS AND CONSTRUCTION	1	
1.1	Location and Governing Law	1	
1.2	Currency	1	
1.3	Currency Conversion	1	
1.4	Defined Terms	2	
-	Construction	9	
	Accounting Terms	10	
1.7	Effective Date	10	
2	APPOINTMENT OF SUPERVISOR	10	
2.1	Appointment	10	
PART	п	11	
CONS	STITUTION AND TERMS OF STOCK	11	
3	CONSTITUTION AND ISSUE	11	
3.1	General	11	
3.2	Terms of Issue	11	
	Redemption at the Option of the Holders of Stock (Change of Control)	11	
	Step-up in Rate of Interest	13	
3.3	Description and Nature	14	
3.4 3.5	Provisions Applicable to Accommodation Stock Ranking of Stock	14 15	
3.6	Validity of Stock	15	
4			
4 RIGH	COVENANT TO PAY: MISCELLANEOUS PROVISIONS AFFECTING HO	LDEKS 15	
4.1	Covenant to Pay	15	
	Pro Tanto Satisfaction	15	
	Holders' Right to Sue	15	
4.4	Terms and Conditions	16	
4.5	Identity of Holders	16	
4.6	Exclusion of Equities	16	
4.7	Unclaimed Money	16	
4.8	Receipt By Holders	17	
4.9	Stock Payable by Instalments	17	
4.10	Purchase, Cancellation, Reissue	18	
4.11	Commissions	18	
5	CERTIFICATES	18	
5.1	Certificates	18	
5.2	Forms of Certificate: Global Certificates	18	
5.3	Execution	19	

5.4	Replacement Certificates	19
5.5	Confirmation	19
6	REGISTERS	19
6.1	Establishment and Maintenance	19
6.2	Registrar	19
6.3	Change of Address	19
6.4	Inspection	20
6.5	Closure	20
6.6	Audit	20
6.7	New Zealand Statutory Requirements	20
7	TRANSFER OF STOCK	21
7.1	Transfer	21
7.2	Death, Insanity or Bankruptcy of Holder	21
7.3	Terms of Stock Prevail	21
PAR	T III	21
CROSS GUARANTEES		21
8	GUARANTEES	21
8.1	Payment	22
8.2	Liability Not Affected	22
8.3	Principal Obligation	22
8.4	Continuing Guarantees	23
8.5 8.6	Discretion as to Enforcement	23 23
8.7	Payment in Gross Claims in Competition	23
8.8	Non-Prejudice	23
8.9	Void or Voidable Payments	24
	Suspension of Rights	24
8.11		24
8.12		24
9	JOINING OF GUARANTEEING SUBSIDIARIES	25
9.1	Joining of Guaranteeing Subsidiaries	25
9.2	Mode of Joining	25
9.3	Release of Guaranteeing Subsidiary etc.	25
9.4	"Material Subsidiary" Defined	25
PAR	ΓΙν	26
COVE	ENANTS	26
10	GENERAL COVENANTS	26
	Authorisations	26
	Maintain Accounts	26
	Attendance at Meetings	26
10.4	Conduct of Business	26

10.5	Compliance with Obligations	26
	Compliance with Legislative Requirements	26
	Provide Offering Documents to Supervisor	27
	Auditors	27
10.9	Notice of Breach	27
11	RANKING OF STOCK AND NEGATIVE PLEDGE	27
	Ranking	27
	Negative Pledge	27
11.3	Charges Permitted	28
PART	V	29
RESO	LUTIONS AND REPORTS FOR SUPERVISOR	29
12	REPORTING	29
12.1	Quarterly Reports	29
12.2	Reports	29
12.3	Financial Statements	29
12.3A	Guaranteeing Group Audit on Request	30
12.4	Auditor's Report	30
	Special Reports	31
	Notices etc	31
	Presentation of Reports	31
12.8	Guaranteeing Group Financial Ratio Requirements	31
13	APPOINTMENT OF AUDITOR	32
13.1	Consultation with Supervisor	32
	Specified Engagement	33
13.3	Terms of Appointment	33
14	MISCELLANEOUS NOTIFICATION/REPORTING REQUIREMENTS	34
14.1	Requested information and reports	34
	Contravention or possible contravention of Issuer Obligations	34
14.3	Serious financial problems	34
PART	VI	35
DEFA	ULT AND ENFORCEMENT	35
15	EVENTS OF DEFAULT: ACCELERATION	35
15.1	Events of Default	35
15.2	Restrictions on Action	36
16	SUPERVISOR'S POWERS OF ENFORCEMENT	36
17	DISTRIBUTION OF FUNDS	37
18	MISCELLANEOUS PROVISIONS AS TO PAYMENT	37
18.1	Payment of Principal Before Interest	37

	Supervisor's Powers to Postpone Distribution	37
	Notice of Distribution	37
18.4	Production of Certificates on Payment by Supervisor	38
PART	VII	38
PROV	38	
19	POWERS OF INVESTMENT	38
20	APPLICATIONS TO COURT	38
21	POWERS OF SUPERVISOR	39
22	SUPPLEMENTAL POWERS OF SUPERVISOR	40
22.1	Reliance on Advice	40
	Resolutions	40
	Application of Funds	40
	Custody of Documents	40
	Defects in Title: Validity of Certificates	41
	Discretion as to Powers	41
	Delegation	41
	Indemnity of Supervisor	41
	Other Dealings	43
	Represent Holders	43
22.11	Power to engage experts	43
23	CHANGE OF SUPERVISOR	44
23.1	Retirement or removal of Supervisor	44
23.2	Requirements for Retirement and Removal	44
23.3	Appointment of Substitute Supervisor	44
24	SUPERVISOR'S POWERS CUMULATIVE	45
25	SUPERVISOR'S REMUNERATION	45
25.1	Remuneration	45
25.2	Expenses	45
25.3	Payable on Demand	45
25.4	Consents etc.	46
PART	VIII	46
MOD	FICATION OF TRUST DEED AND SUBSTITUTION	46
26	MODIFICATIONS	46
26.1	Amendments affecting Retail Stock	46
26.2	Amendments affecting Wholesale Stock	47
27	WAIVER	48
27.1	Temporary Variation	48

27.2	Waivers	48	
28 EXEM	CONSTRUCTION BY REFERENCE TO ANALOGOUS OBLIGATIONS AND PTIONS	48	
29	SUBSTITUTED OBLIGOR	48	
29.1	Substitution	48	
29.2	Release of Obligations	49	
	Amendments	49	
29.4	Solvency	50	
	Interests of Holders	50	
	Supervisor's Discretion	50	
PART	IX	50	
COUP	COUPONHOLDERS		
30	COUPONS	50	
30.1	General	50	
30.2	Notices	51	
30.3	Assumption as to Identity of Couponholders	51	
PART	x	51	
MISC	MISCELLANEOUS PROVISIONS		
31	NOTICES	51	
31.1	Notices	51	
31.2	Effectiveness	52	
31.3	Proof of Notice	52	
31.4	Bearer Stock	53	
32	MEETINGS OF HOLDERS	53	
32.1	Convened by Spark NZ	53	
32.2	Convened by the Supervisor	53	
32.3	Other Rules	53	
33	RELEASE UPON PAYMENT	53	
FIRST SCHEDULE			
SECOND SCHEDULE			
THIRD SCHEDULE			
FOURTH SCHEDULE			

TRUST DEED RELATING TO THE CONSTITUTION AND ISSUE OF UNSECURED STOCK

THIS TRUST DEED is made as of the 25th day of October 1988, as amended and restated on 19 September 2012, 18 September 2014 and by the amending and restating deed dated 4 November 2015.

PARTIES

Spark New Zealand Limited (formerly Telecom Corporation of New Zealand Limited), a company duly registered under the Companies Act 1993 (*Spark NZ*)

The New Zealand Guardian Trust Company Limited, a company incorporated in New Zealand and having its registered office at Auckland (*Supervisor*)

BACKGROUND

- A Spark NZ and/or any Guaranteeing Subsidiary may from time to time accept loans or deposits from, or issue bonds or other debt securities to, any Person or Persons.
- B The Directors have resolved to enter into this trust deed to provide for the constitution and issue of Stock (including Bonds) in respect of certain indebtedness of the nature referred to in Recital A.
- C The Supervisor has agreed for the consideration hereinafter expressed to act as Supervisor for the benefit of the Holders (as herein defined) upon and subject to the terms and conditions of, and with the powers and authorities contained in, this Deed.

NOW THIS DEED WITNESSES AND IT IS HEREBY AGREED AND DECLARED as follows:

PART I

INTERPRETATION AND APPOINTMENT OF SUPERVISOR

1 DEFINITIONS AND CONSTRUCTION

1.1 Location and Governing Law

This Deed shall be construed and take effect as a contract and declaration of trust made in New Zealand and shall be governed by and construed in accordance with the law of New Zealand.

1.2 Currency

All references to money in this Deed shall, unless the contrary is stated or necessarily implied, be deemed to be references to New Zealand currency.

1.3 Currency Conversion

Where for the purposes of any Stock or any provision of this Deed it is necessary to determine the New Zealand currency equivalent of a sum expressed in a foreign currency such sum shall for the purposes of this Deed, unless otherwise agreed in writing by the Supervisor either generally or in any particular case, be converted to

New Zealand currency on the basis of conversion employed in preparing the Latest Balance Sheet or such other basis as is agreed from time to time with the Auditors, applied at or about the close of business in New Zealand on the day on which such calculation falls to be made, and any Stock which is payable in a foreign currency shall for the purposes of this Deed (but without affecting the nature or amount of the indebtedness of the Guaranteeing Group in respect of any such Stock or the terms and conditions of any Stock prescribing an alternative method of conversion) be treated as being of an amount equal to the New Zealand currency equivalent determined as aforesaid, provided always that in determining the New Zealand currency equivalent of any asset or liability in a foreign currency, account may be taken of any contract or arrangement in force for covering the risk of fluctuations between New Zealand currency and the foreign currency of the asset or liability (as the case may be).

1.4 **Defined Terms**

In this Deed and the Schedules hereto unless the context otherwise requires:

1st Rating Change has the meaning set out in Clause 3.2B;

2nd Rating Change has the meaning set out in Clause 3.2B;

Assets means, in respect of any corporation, all or any part of the present and future undertaking, property, assets and revenues (including any right to receive revenues) of such corporation whatsoever and wheresoever and its issued but uncalled and called but unpaid capital;

Accommodation Stock means Stock issued in accordance with Clause 3.3(b);

Auditors means at any time the Person holding office as Spark NZ's auditor or (in the event of that Person being unable or unwilling to carry out any action required of him pursuant to the Deed) such firm of accountants as may be nominated or approved by the Supervisor in consultation with Spark NZ;

Balance Sheet means (in relation to any of the Consolidated Group Members) a statement of financial position and statement of financial performance of the relevant Consolidated Group Member or (in relation to any group of Consolidated Group Members) a consolidated statement of financial position and statement of financial performance of the relevant group, prepared as at the end of its financial year or as at any other date (as the case shall be or the context shall require) in a manner that complies with the FMC Act which balance sheet if prepared at the end of the financial year has been audited by the Auditor and is accompanied by his report;

Bearer Bonds or *Bearer Stock* means Bonds or Stock represented by Certificates issued to bearer;

Bonds means Stock constituted by this Deed and for the time being outstanding in the form of bonds, and where the context permits or requires, refers also to the Principal represented by any such bonds;

Business Day means any day on which banks are open for the transaction of general banking business in Wellington, New Zealand and (but only as regards the provisions of this Deed relating to the time or manner of payments required to be made in respect of any Stock) in any other jurisdiction in which such payment is to be made pursuant to the terms of such Stock;

Certificate means, in respect of any Stock, any certificate, acknowledgement, receipt or other written evidence or indication issued by or on behalf of the Issuer thereof representing, or evidencing the issue of, that Stock, and which (without limitation) may comprise (i) a form of bond, note or other instrument or security; or (ii) a notice or acknowledgement of registration or entitlement;

Change of Control has the meaning set out in Clause 3.2A;

Change of Control Period has the meaning set out in Clause 3.2A;

Change of Control Put Notice has the meaning set out in Clause 3.2A;

Charge means any lien, pledge, charge, mortgage, hypothecation or other encumbrance or security interest, but does not include:

- (i) leases, bailments or reservations of title; or
- (ii) the rights of resumption conferred by Sections 27 to 27D of the State-Owned Enterprises Act (as inserted by the Treaty of Waitangi (State Enterprise Act 1988)) or any memorial recording such rights;

Class means:

- (i) in relation to Stock, each category of Stock which the Supervisor determines on a reasonable basis should be treated as constituting a separate Class of Stock;
- (ii) in relation to Holders, the Holders of each such separate Class of Stock;

Companies Act means the Companies Act 1993;

Consolidated Group means Spark NZ and those entities which are consolidated with Spark NZ (including "in substance subsidiaries") for the purpose of the preparation of consolidated financial statements in compliance with the FMC Act, including NZ GAAP, and *Consolidated Group Member* means any one of them;

Coupon means a bearer coupon appertaining to any Bearer Stock and includes any replacement coupon;

Couponholders means the several persons who are for the time being the holders of Coupons;

debt security has the meaning ascribed thereto in the FMC Act;

Director means a director of Spark NZ (or, as the context may otherwise expressly or by necessary implication require, any Guaranteeing Subsidiary) for the time being, and includes an alternate director acting as a director of Spark NZ (or any Guaranteeing Subsidiary) and *Directors* means the directors (including any committee thereof) acting in accordance with the constitution of Spark NZ (or any Guaranteeing Subsidiary) for the time being;

EBIT means, in relation to the Consolidated Group or the Guaranteeing Group and any financial period, as the case may be, the aggregate amount on a consolidated basis of the earnings of the relevant group before charging interest and before charging or providing for income tax or any other tax, in each case in relation to that financial period and calculated in accordance with the FMC Act, including NZ GAAP, and excludes:

- profits and losses derived on the sale of fixed assets or investments or revaluations;
- (ii) non-cash equity accounted profits and losses; and
- (iii) unrealised gains and losses resulting from foreign exchange, swaps, futures, options or similar transactions;

Effective Date has the meaning set out in Clause 1.7;

Event of Default means, subject to Clause 15.2, any of the events or circumstances set out in Clause 15.1;

Financial Reporting Act means the Financial Reporting Act 2013;

FMC Act means the Financial Markets Conduct Act 2013;

FMC Regulations means the Financial Markets Conduct Regulations 2014;

Global Certificate means a Certificate for an amount of Stock (whether in registered or bearer form) which:

- evidences or represents a global, aggregate or composite entitlement to, and/or may be divisible into, separate component parcels of such Stock; and/or
- (ii) against which interests or entitlements or dealings in respect of separate component parcels of such Stock may be separately transacted, registered, noted or recorded (whether through operation of any clearing system or otherwise); and/or
- (iii) which may be exchangeable in whole or in part for Certificates in definitive form representing separate component parcels of such Stock;

Guaranteeing Group means Spark NZ and the Guaranteeing Subsidiaries, and *Guaranteeing Group Member* means any of them;

Guaranteeing Subsidiary means at any time and from time to time any Subsidiary which has become a Guaranteeing Subsidiary in accordance with Clause 9, but does not include any Subsidiary which has been released from its obligations as a Guaranteeing Subsidiary in accordance with Clause 9, and *Guaranteeing Subsidiaries* shall be construed accordingly;

Holders means:

- (i) in respect of Registered Stock, the several Persons for the time being entered in the relevant Register as the holders of such Stock; and
- (ii) in the case of Bearer Stock, the several Persons for the time being holding Certificates in respect of such Stock,

and includes their personal representatives;

Issuer means, in relation to any Stock, the relevant Guaranteeing Group Member which is the issuer of such Stock;

Issuer Obligations has the meaning given to it in the FMC Act;

in writing and *written* includes words written, printed, typewritten, lithographed, telexed, cabled or otherwise represented or reproduced in visible form by any other means;

Latest Financial Statements means, at any time and from time to time, in relation to any particular Consolidated Group Member or group of Consolidated Group Members, the then most recent periodic statement of financial position and statement of financial performance for that Consolidated Group Member or group (and, if for a group, on a consolidated basis) as at and for the period ending on the relevant date, that has been prepared and presented to the Directors, whether such comprises the Latest Balance Sheet or other audited accounts or unaudited management accounts;

Latest Balance Sheet means at any time the then most recent Balance Sheet which has been prepared in respect of the relevant Consolidated Group Member or group of Consolidated Group Members;

a reference to something having a *material adverse effect* on a Guaranteeing Group Member is a reference to it having a material adverse effect on:

- that Person's ability to perform or comply in any material respect with its obligations under this Deed or on any Stock issued by it or on which it is liable;
- (ii) the financial condition of the Guaranteeing Group (taken as a whole);

Materiality Certificate means, where such phrase is employed in this Deed in relation to certain circumstances or occurrences and/or the Supervisor's rights or powers in respect thereof, a certificate given by the Supervisor in relation to those

circumstances or occurrences stating that in its opinion (which shall for the purposes of this Deed be final and conclusive) such circumstances or occurrences are or are likely to become materially prejudicial to the general interests of Holders and justify the relevant action taken or omitted or proposed to be taken or omitted, or the relevant request made or proposed to be made by the Supervisor in respect thereof pursuant to the relevant provision(s) of this Deed;

Material Subsidiary has the meaning set out in Clause 9.4;

Moody's has the meaning set out in Clause 3.2A;

Non-Guaranteeing Subsidiaries means any Subsidiaries of Spark NZ which are not Guaranteeing Subsidiaries;

NZ GAAP means generally accepted accounting practice as defined in section 6 of the FMC Act;

Offer Document means the product disclosure statement, or other disclosure document required by the FMC Act, relating to the issue of Stock, in each case which has been prepared by, or on behalf of, the Issuer, and shall include (in each case) all documents to be distributed with or which form part of the relevant documents;

outstanding means, in relation to Stock, all the Stock issued other than:

- (i) any Stock which has been repaid or redeemed pursuant to its terms and conditions; or
- (ii) any Stock in respect of which the date for repayment or redemption pursuant to its terms and conditions has occurred and the repayment or redemption moneys for which (including all interest accrued thereon to the date for such repayment or redemption) have been duly paid to or to the order of the Supervisor or to any relevant paying agent and remain available for payment against presentation of Stock and/or Coupons as the case may be; or
- (iii) any Stock which has become void pursuant to its terms and conditions;
- (iv) any Stock which has been purchased and cancelled pursuant to its terms and conditions,

and (for certainty) it is declared that where any Certificate for Stock has been replaced in whole or in part by new Certificates issued in respect of such Stock then, to the extent so replaced and without prejudice to the Stock evidenced by such new Certificates, the original Certificate shall not represent outstanding Stock;

Person includes an individual, a corporation sole, a company or any other body corporate, any association of Persons whether corporate or unincorporate, any state or government or department or agency thereof, and any legislative body authority or agency whether having separate legal personality or not;

Principal means:

- (i) in relation to any Stock other than Accommodation Stock, the amount (other than interest) payable on the redemption or repayment of such Stock inclusive of the premium (if any) payable on redemption or repayment in accordance with the conditions of issue thereof; and
- (ii) in relation to Accommodation Stock means the aggregate amount of all indebtedness (other than interest) from time to time owing or outstanding to the Holder as is described in the Certificate for such Stock;

Put Date has the meaning set out in Clause 3.2A;

Put Event has the meaning set out in Clause 3.2A;

Put Event Notice has the meaning set out in Clause 3.2A;

Put Period has the meaning set out in Clause 3.2A;

rating agency has the meaning set out in Clause 3.2A;

Rating Change has the meaning set out in Clause 3.2B;

Register means the register or registers of Stock kept pursuant to Clause 6 of this Deed or any one or more of them as the case may be;

Registered Holder means a Holder of Registered Stock;

Registered Address in respect of a Registered Holder means its address for the time being recorded in the Register;

Registered Stock means all Stock (other than Bearer Stock) held by Persons whose names are from time to time entered in the Register;

Registrar means, in respect of any Stock, the Person (whether it be the Issuer of such Stock or any other Person appointed by it for the purpose) which maintains the Register in respect of such Stock);

Relevant Announcement Date has the meaning set out in Clause 3.2A;

Relevant Person has the meaning set out in Clause 3.2A;

Retail Holders means a Holder of Retail Stock;

Retail Stock means Stock offered or sold as a regulated offer under the FMC Act or offered in accordance with clause 19 of schedule 1 of the FMC Act or any Stock that is designated as Retail Stock in the supplement to this Deed which constitutes the terms of the Stock;

S&P has the meaning set out in Clause 3.2A;

Special Resolution means a Special Resolution of the Holders or any Class of Holders (as the case may require) as defined in regulation 14.1 of the Fourth Schedule;

State-Owned Enterprises Act means the State-Owned Enterprises Act 1986;

Stock means any and all unsecured indebtedness, liabilities, obligations or debt securities (including without limitation Bonds and Accommodation Stock) of or issued by any Guaranteeing Group Member which is or are issued as Stock constituted by and with the benefit of this Deed from time to time outstanding, or a specified portion thereof as the case may require, and includes the Principal represented thereby;

Stock Moneys means at any time and in relation to the Stock all moneys payable on or in relation to the Stock to the relevant Holders pursuant to the terms and conditions thereof or of this Deed;

Subsidiary means any corporation which is a subsidiary of Spark NZ in terms of sections 5 to 8 (inclusive) of the Companies Act;

Substitute Rating Agency has the meaning set out in Clause 3.2A;

Supervisor means The New Zealand Guardian Trust Company Limited or any other or substitute supervisor or supervisors for the time being hereof;

Total Amount in relation to Accommodation Stock means the total amount owing or outstanding in respect of all indebtedness liabilities or obligations covered by such Stock as referred to in the Certificate therefor;

the Trust Deed and this Deed means this Trust Deed (including the Schedules hereto) as originally executed and as it may from time to time be supplemented, modified or novated in accordance with the provisions in that behalf hereinafter contained, and when the context so admits includes any deed which is or is expressed to be supplemental to or collateral with this Deed and in particular shall include supplemental deeds executed by any of the Guaranteeing Subsidiaries in accordance with Clause 9;

Tangible Assets means all Assets that would be regarded as being tangible assets in accordance with the accounting principles and practices adopted in Spark NZ's Latest Balance Sheet with only such changes (if any) thereto as may have been approved by the Auditors;

Total Tangible Assets means, in respect of any particular Consolidated Group Member or group of Consolidated Group Members, the aggregate of the book values (determined in accordance with the principles and practices applied in the Latest Balance Sheet with only such changes (if any) thereto as may have been approved by the Auditors) of the Tangible Assets of such Consolidated Group Member or group (and, if for a group, on a consolidated basis) as at any time and from time to time valued and disclosed in the Latest Financial Statements of such Consolidated Group Member or group, and for the purposes of this definition the Supervisor shall accept a certificate of any two Directors on behalf of them all as to any relevant matter that has occurred since the date of the Latest Balance Sheet of that Consolidated Group Member or group.

Wholesale Holder means a Holder of Wholesale Stock.

Wholesale Stock means Stock which is not permitted to be offered or sold to members of the public for the purposes of the Securities Act 1978 or to retail investors for the purpose of the FMC Act or any Stock that is designated as Wholesale Stock in the supplement to this Deed which constitutes the terms of the Stock and which the relevant Holder or Holders subscribed or otherwise acquired such Stock in compliance with such restrictions;

1.5 Construction

In the construction of the Deed:

- (a) *Headings*: The headings and marginal notes appear as a matter of convenience and shall not affect the construction of this Deed;
- (b) *Clauses*: In the absence of an express indication to the contrary, references to Parts, Clauses, sub-clauses, paragraphs and Schedules are to the parts, clauses, sub-clauses and paragraphs of, and schedules to, this Deed;
- (c) Statutes: References to any statute, statutory regulations or other statutory instrument shall be deemed to be references to the statute, statutory regulations or instrument as from time to time amended or re-enacted or, as the context permits, provisions substituted therefor for the time being in force, and in addition reference to any statute includes any statutory regulations or instruments from time to time made and in force under that statute;
- (d) *Agreements*: References to any deed agreement or other instrument shall be read as referring to such deed agreement or other instrument as from time to time modified supplemented or novated;
- (e) Statutory Definitions: Except where inconsistent with specific definitions thereof contained herein, words defined in the Companies Act, the FMC Act or the Financial Reporting Act shall have the same meanings in this Deed; and in the case of conflict the definitions in the FMC Act shall prevail over those in either of the other Acts and the definitions in the Financial Reporting Act shall prevail over those in the Companies Act;
- (f) *Singular*: The singular includes the plural and vice versa, and words importing any gender include the other genders;
- (g) *Joint and Several*: All obligations of Spark NZ and the Guaranteeing Subsidiaries under this Deed are joint and several;
- (h) *Successors and Assigns*: Each reference to a party to this Deed shall be read as a reference to that party and its successors and assigns.

1.6 Accounting Terms

Expressions that are utilised in connection with accounting functions or reporting or in the description of either thereof in this Deed shall bear the respective meanings accepted in respect of or ascribed to them in the preparation of the Latest Balance Sheet of Spark NZ.

1.7 Effective Date

This Deed shall become effective on the date (the *Effective Date*) specified in that regard by not less than 3 days prior notice in writing from Spark NZ to the Supervisor. Until the Effective Date has occurred the provisions of this Deed shall not be effective and Spark NZ shall not issue any securities which are or purport to be Stock constituted by this Deed.

2 **APPOINTMENT OF SUPERVISOR**

2.1 Appointment

Spark NZ hereby appoints the Supervisor, and the Supervisor hereby accepts appointment, as supervisor and trustee for the Holders in respect of the Stock on the terms and subject to the provisions of this Deed and applicable law, and with the rights, powers, duties and obligations in respect of any Stock conferred by this Deed including, without limitation:

- (a) acting on behalf of the Holders in relation to:
 - (i) the Issuer;
 - (ii) any matter connected with this Deed; and
 - (iii) any contravention or alleged contravention of the Issuer Obligations; and
- (b) supervising the Issuer's performance:
 - (i) of its Issuer Obligations; and
 - (ii) in order to ascertain whether the assets of the Issuer are or may be available, whether by way of security or otherwise, are sufficient or likely to be sufficient to discharge the Principal, interest and other moneys payable on, or in relation to the Stock as they become due; and
- (c) performing or exercising any other functions, duties, and powers conferred or imposed on the Supervisor by or under the FMC Act, the Financial Markets Supervisors Act 2011 or this Deed.

2.2 Hold in trust

The Supervisor shall hold the following in trust for the benefit of the Holders:

(a) the right to enforce the Issuer's duty to repay, or to pay interest, under the terms of the Stock;

- (b) any charge or security for repayment; and
- (c) the right to enforce any other duties that any Guaranteeing Group Member and any other person have under:
 - (i) the terms of the Stock; or
 - (ii) the provisions of this Deed or the FMC Act in relation to the Stock.

PART II

CONSTITUTION AND TERMS OF STOCK

3 CONSTITUTION AND ISSUE

3.1 General

Stock may be issued by any Guaranteeing Group Member as Issuer at any time and from time to time to such Persons, in such amounts, in such currencies and upon such terms and conditions (not being inconsistent with the provisions contained in this Deed) as it may from time to time determine.

3.2 Terms of Issue

Without limiting the generality of Clause 3.1, Stock may be issued either at par or at a premium or at a discount, and bearing such respective rates of interest (whether fixed variable or determined by a stated method) or without interest, and payable or redeemable at such respective times or periods or on demand and at par or at a discount or with payment of such premiums or allowances (whether fixed variable or determined by a stated method) or on conversion into debt securities of any Consolidated Group Member, and whether securing or evidencing indebtedness, liabilities or obligations of the Issuer for borrowed money or securing or evidencing other obligations or debt securities of the Issuer, in all cases as the Issuer may from time to time determine.

3.2A Redemption at the Option of the Holders of Stock (Change of Control)

(A) A "Put Event" will be deemed to occur if:

- (i) any person or persons acting in concert (within the meaning of Rule 6(2)(a) of the New Zealand Takeovers Code) (the "Relevant Person") at any time directly or indirectly acquires: (A) more than 50 per cent of the issued ordinary share capital of Spark NZ; or (B) such number of shares in the capital of Spark NZ carrying more than 50 per cent of the total voting rights attached to the issued share capital of Spark NZ that are normally exercisable at a general meeting of Spark NZ (such event being a "Change of Control"); and
- (ii) on the date (the "Relevant Announcement Date") of the first public announcement of the relevant Change of Control, the Stock carries from either Moody's Investors Service Limited ("Moody's") or Standard & Poor's Rating Services, a division of the McGraw-Hill Companies, Inc. ("S&P") or any of their respective successors or any other rating

agency (each a "Substitute Rating Agency") of equivalent international standing from whom a rating is procured by Spark NZ (each a "rating agency"):

- (1) an investment grade rating (*Baa3/BBB-, or equivalent, or better*), and such rating from any rating agency is, within the Change of Control Period, either downgraded to a non-investment grade credit rating (*Ba1/BB+, or equivalent, or worse*) or withdrawn and is not, within the Change of Control Period, subsequently (in the case of a downgrade) upgraded or (in the case of a withdrawal) reinstated to an investment grade credit rating by such rating agency; or
- a non-investment grade credit rating (*Ba1/BB+, or equivalent, or worse*) or no credit rating, and no rating agency assigns, within the Change of Control Period, an investment grade credit to the Stock,

provided that if on the Relevant Announcement Date the Stock carries a credit rating from more than one rating agency, at least one of which is investment grade, then sub-paragraph (1) will apply.

- (B) If a Put Event occurs, the Holder of any Stock will have the option to require the Issuer to purchase (or procure the purchase of) such Stock on the date which is seven days after the expiration of the Put Period (the "Put Date") at 101% of the scheduled redemption amount of the Stock, together with interest accrued to but excluding the Put Date.
- (C) Upon Spark NZ becoming aware that a Put Event has occurred, Spark NZ shall immediately notify the Supervisor and shall, and at any time upon the Supervisor becoming similarly so aware the Supervisor shall require Spark NZ to, give notice (a "Put Event Notice") to the Holders in accordance with Clause 31 specifying the nature of the Put Event and the procedure for exercising the option contained in this Clause 3.2A.
- (D) To exercise the option to require the purchase of a Stock under this Clause 3.2A, a Holder of Stock must, at the specified office of Spark NZ, at any time during normal business hours of Spark NZ falling within the period (the "Put Period") of 60 days after a Put Event Notice is given, deliver a duly signed and completed notice of exercise in the form (for the time being current) obtainable from the specified office of Spark NZ (a "Change of Control Put Notice"). A Change of Control Put Notice, once given, shall be irrevocable. The Issuer shall purchase (or procure the purchase of) the Stock specified in the relevant Change of Control Put Notice on the Put Date unless previously redeemed and cancelled or purchased. Payment in respect of any Stock so purchased will be made on the Put Date by transfer to the bank account of the relevant Holder specified in the relevant Change of Control Put Notice.
- (E) If the rating designations employed by either Moody's or S&P are changed from those which are described in paragraph (ii) of the definition of "Put

Event" above, or if a rating is procured from a Substitute Rating Agency, Spark NZ shall determine, with the agreement of the Supervisor (in respect of which the Supervisor shall be entitled to obtain and rely upon any expert advice pursuant to Clause 22.1), the rating designations of Moody's or S&P or such Substitute Rating Agency (as appropriate) as are most equivalent to the prior rating designations of Moody's or S&P and this Clause 3.2A shall be read accordingly.

- (F) The Supervisor is under no obligation to ascertain whether a Put Event or Change of Control or any event which could lead to the occurrence of, or could constitute, a Put Event or Change of Control has occurred and, until it shall have actual knowledge or notice thereof to the contrary, the Supervisor may assume that no Put Event or Change of Control or such other event has occurred.
- (G) In this Clause 3.2A, "Change of Control Period" means the period commencing on the Relevant Announcement Date and ending 180 days after the Relevant Announcement Date.

3.28 Step-up in Rate of Interest

- (A) With effect from the date of each Rating Change, the interest rate applicable to each Class of Stock (other than zero coupon Stock) shall be adjusted to be equal to:
 - (i) in the case of a 1st Rating Change, the aggregate of the interest rate applicable to the Class of Stock as at 30 November 2011 plus 0.50% per annum; and
 - (ii) in the case of a 2nd Rating Change, the aggregate of the interest rate applicable to the Class of Stock as at 30 November 2011 plus 1.00% per annum.
- (B) If, in relation to a Class of Stock, a Rating Change occurs other than on a date on which interest is scheduled to be paid on the Class of Stock, the amount of interest that is payable on the next scheduled interest payment date for the Class of Stock shall be the amount reasonably determined by Spark NZ by reference to the different rates of interest that are applicable to the Class of Stock for each period from (and including) the immediately preceding interest payment date to (but excluding) the immediately succeeding interest payment date during which a different interest rate is applicable to the Class of Stock in accordance with this Clause 3.2B.
- (C) If a Rating Change occurs prior to the scheduled redemption date of any Class of zero coupon Stock, the scheduled redemption amount of the Class of zero coupon Stock shall be adjusted by an amount reasonably determined by Spark NZ that represents the equivalent economic effect of the increase in the interest rate that applies to Stock that bears interest.
- (D) Spark NZ shall notify the Supervisor and the Holders in accordance with Clause 31 of the adjustment to the interest rate and the amount of interest

payable to the Holders following that adjustment (or the adjustment to the scheduled redemption amount in the case of zero coupon Stock) as soon as reasonably practicable following each Rating Change.

- (E) Any reasonable determination by Spark NZ under this Clause 3.2B shall be final and (in the absence of manifest error) binding on all Holders.
- (F) This Clause 3.2B applies only to Stock that is outstanding as at 30 November 2011.
- (G) In this Clause 3.2B:

"1st Rating Change" means the first date on which the Stock carries a credit rating of BBB+ by S&P or Baa1 by Moody's or the equivalent by a Substitute Rating Agency.

"2nd Rating Change" means the first date on which the Stock carries a credit rating of BBB or worse by S&P or Baa2 or worse by Moody's or the equivalent by a Substitute Rating Agency.

"Rating Change" means a 1st Rating Change or a 2nd Rating Change.

3.3 Description and Nature

Without limiting the generality of Clause 3.1, Stock may be issued in the form of or described or named as loan stock, debentures, bonds, notes, bills, savings, deposits or in such other form or under such other name or description as the Issuer shall determine, and:

- (a) *Bonds*: It is acknowledged that Spark NZ intends following execution of this Deed and from time to time to issue Stock in the form of Bonds; and
- (b) Accommodation Stock: Stock may from time to time be issued as Accommodation Stock as cover for and in order to render applicable the provisions of this Deed to any indebtedness, liabilities or obligations of any Guaranteeing Group Member to any Person, whether generally or in respect of any particular or particular class of arrangement or financial accommodation. For the purposes of this Deed "Accommodation Stock" shall be Stock issued under a Certificate describing it as such (which may be in the form set out the First Schedule) and describing the nature of the indebtedness, liabilities or obligations which such Stock is to cover.

3.4 **Provisions Applicable to Accommodation Stock**

Accommodation Stock shall for all purposes and on all occasions be deemed to be validly issued to each Holder thereof as cover for the indebtedness, liabilities and obligations expressed in the relevant Certificate to be covered by the Accommodation Stock so held and interest thereon. Notwithstanding that the amount owing to the Holder thereof may at any time be less than the nominal amount of the Accommodation Stock so held, such Holder shall not on that account be liable to make any payment to the Guaranteeing Group or to the Supervisor in respect of the Accommodation Stock so held or be obliged to make any further advance or afford any further accommodation to the Guaranteeing Group except as may expressly be agreed between the Guaranteeing Group and such Holder. The nominal amount for which Accommodation Stock is issued shall be determinative of the Holder's right to vote and (except as otherwise expressly provided in this Deed) in all matters under this Deed, other than the right to participate pari passu with other Holders in distributions pursuant to the provisions of Clause 17 which shall be determined on the Total Amount of the Accommodation Stock as described in the Certificate therefor.

3.5 Ranking of Stock

The Stock shall rank inter se pari passu as to payment of the Stock Moneys evidenced thereby notwithstanding that it may be issued at different times or on different terms or in different forms or denominated in different currencies.

3.6 Validity of Stock

No Person holding any Certificate shall be concerned or obliged to enquire whether any Stock purporting to be evidenced by it has been issued in contravention of any provision of this Deed. All Stock issued for consideration, the Certificate for which indicates, or the entry on the appropriate Register for which indicates, or in respect of which other evidence indicates, that it has ostensibly been issued or was intended to be issued as Stock pursuant to the provisions of this Deed shall be validly issued and constituted by this Deed and entitled to the benefit of the provisions of this Deed notwithstanding that the issue of that Stock or any related security was in breach of any provision of this Deed. This clause shall not however prejudice or affect the Supervisor's rights under or pursuant to this Deed against the Guaranteeing Group in relation to such breach.

4 COVENANT TO PAY: MISCELLANEOUS PROVISIONS AFFECTING HOLDERS' RIGHTS

4.1 **Covenant to Pay**

Each Issuer covenants with the Supervisor that it will pay the Stock Moneys of the Stock issued by it as and when due and payable in accordance with the relative terms and conditions of the Stock to or to the order of the Supervisor in the currency and manner in which the same is payable.

4.2 **Pro Tanto Satisfaction**

Notwithstanding Clause 4.1 every payment to Holders of or on account of the Stock Moneys due and payable on the Stock in accordance with its relative terms and conditions shall be in satisfaction pro tanto of the indebtedness acknowledged and covenanted to be paid in Clause 4.1.

4.3 Holders' Right to Sue

No Holder shall be entitled to proceed:

 (a) directly against any Guaranteeing Group Member for the enforcement, observance or performance of any term or condition of the Stock held by such Holder or of this Deed unless the Supervisor fails to so proceed having become bound to do so and such failure shall continue; (b) either against the Guaranteeing Group or against the Supervisor, for the enforcement, observance or performance of any provision of this Deed that is solely for the benefit of the Supervisor.

4.4 **Terms and Conditions**

The Stock shall be held by the Holders with the benefit of and subject to the following provisions, terms and conditions (referred to in this Deed as the terms and conditions of or applicable to the relevant Stock):

- (a) such of the provisions of this Deed as are applicable thereto having regard to the Class of such Stock; and
- (b) the terms and conditions endorsed on or relating to the Certificates therefor and the terms and conditions specified to be applicable to such Stock in any Offer Document or other document (whether supplemental to this Deed or not) issued or executed by the Issuer in relation to the issue, allotment or offering of such Stock;

and all such provisions terms and conditions shall be binding upon the Issuer and the Holder and all Persons claiming under or through them respectively. The Holders and all Persons claiming by or through them are deemed to have notice of all such provisions, terms and conditions.

4.5 Identity of Holders

The Holders are to be regarded as the beneficial owners of the Registered Stock registered in their names respectively, and of the Bearer Stock in respect of which they hold the relevant Certificate(s), and are to be regarded as exclusively entitled thereto and the Guaranteeing Group and all other Persons may act accordingly. The Guaranteeing Group shall not be bound to enter in the Register notice of any trust or, save as provided in the relevant Certificate or except as ordered by a Court of competent jurisdiction, to recognise any trust or equity affecting the ownership of any Stock or the money thereby represented.

4.6 Exclusion of Equities

Every Holder will be recognised by the Guaranteeing Group as entitled to the Stock held by it and to the Stock Moneys payable thereon free from any equity set-off or cross claim between the Guaranteeing Group and any prior holder thereof.

4.7 Unclaimed Money

Subject to the terms and conditions of the relevant Stock, any payment of any of the Stock Moneys:

- (a) sent or made by or on behalf of the Guaranteeing Group to a Registered Holder at its last Registered Address that shall be returned unclaimed; or
- (b) due to a Registered Holder whose address is unknown; or
- (c) payable in respect of any Bearer Stock and unclaimed six months after its due date,

shall be either retained by the Guaranteeing Group for such Holder or (if held by or on behalf of the Supervisor) paid by or on behalf of the Supervisor to the Guaranteeing Group (without prejudice to the rights of the Person entitled thereto in priority to the Guaranteeing Group) and, in either case, retained by the Guaranteeing Group for such Holder until claimed without liability to invest the same or pay interest thereon. Any such amount for which payment is not claimed within five years from its due date shall be forfeited by the Directors of the Issuer for its benefit. After forfeiture of any such amount the Person who would otherwise have been entitled thereto shall become entitled to payment of a sum equal to the amount forfeited upon adducing evidence sufficient to satisfy the Directors that such Person would have been entitled thereto had the same not been forfeited.

4.8 Receipt By Holders

The receipt of the Holder, or in the case of joint Holders the receipt of any one of them, for the Stock Moneys from time to time accruing due in respect of any Stock shall be a good discharge to the Guaranteeing Group notwithstanding any notice (express or implied) any Guaranteeing Group Member may have of the right, title, interest or claim of any other Person to or in such Stock or the Stock Moneys payable in respect thereof.

4.9 Stock Payable by Instalments

If by the terms of issue thereof any of the Stock is to be paid for by instalments and the amount of any instalment is not duly paid to the Issuer on due date, the Issuer at any time thereafter before such instalment is paid but only after giving not less than fourteen (14) days' notice of its intention so to do may (subject to any provisions to the contrary in the terms and conditions of such Stock) either:

- (a) forfeit all instalments previously paid and cancel the allotment of the Stock in question; or
- (b) charge and sue for the recovery of such unpaid instalment and charge and sue for the payment of interest thereon, at a rate equivalent to the cost to the Issuer (as determined by the Issuer in its absolute discretion, its certification thereof to be conclusive in the absence of manifest error) of obtaining funds in an amount equivalent to the amount so unpaid, from the date that such instalment became due until the said instalment is paid.

A statutory declaration in writing that the declarant is a Director of the Issuer and that Stock has been duly forfeited on the date stated in the declaration shall be conclusive evidence of the facts therein stated as against all persons claiming to be entitled to such Stock. The Issuer may sell or dispose of such Stock, may receive the consideration if any given for such Stock on any sale or disposition thereof, and may execute a transfer of such Stock in favour of or otherwise deliver or transfer such Stock to the Person to whom such Stock is sold or disposed of and such Person shall thereupon become and (as regards Registered Stock) be registered as the Holder of such Stock and shall not be bound to see to the application of the purchase money if any nor shall such Person's title to such Stock be affected by any irregularity or invalidity in the proceedings in relation to the forfeiture sale or disposal of such Stock.

4.10 Purchase, Cancellation, Reissue

Subject to any provisions to the contrary in the terms and conditions of the Stock:

- (a) *Purchase*: The Issuer or any other Consolidated Group Member may purchase Stock from any Person at any time on any market at any price; and
- (b) *Cancel or Reissue*: The Issuer may cancel any Stock so purchased and/or may re-issue any such Stock, but shall not be obliged to do so.

4.11 Commissions

The Guaranteeing Group shall be at liberty to pay a commission, procuration, fee or brokerage to any Person for subscribing, underwriting the subscription of, or obtaining subscriptions for, any Stock.

5 CERTIFICATES

5.1 Certificates

Subject to the terms and conditions of the relevant Stock, there shall be issued by or on behalf of the Issuer to every Holder a Certificate evidencing the relevant Stock held by it in such denominations and in such form and manner (not being inconsistent with the provisions of this Deed or any applicable law) as shall be determined by the Directors.

5.2 Forms of Certificate: Global Certificates

Without limiting the generality of Clause 5.1 and the ability to issue Certificates in any other form in compliance therewith:

- (a) Certificates for Accommodation Stock may be in the form set out in the First Schedule;
- (b) Certificates may be in the form of computer-generated records, receipts, statements or acknowledgements;
- (c) Certificates may be issued in the form of Global Certificates (whether temporary or not) or of such other nature or form as is necessary or expedient to meet the requirements of the terms and conditions of such Stock, the method or circumstances (including market) of the offer issue or allotment thereof, or the requirements of any applicable clearing system or secondary market in which such Stock may be traded.

The Supervisor shall be supplied with a copy of any form of Certificate (other than a Certificate for Accommodation Stock issued in the form set out in the First Schedule or a Certificate the form of which has previously been supplied under this clause) as soon as practicable following the first issue of a Certificate in that form. Sufficient details shall be entered in the relevant Register as are necessary to identify in respect of all Stock entered thereon the respective forms of Certificate therefor if Stock represented by different forms of Certificate is entered thereon.

5.3 Execution

Certificates may be executed by or on behalf of the Issuer (including by the Registrar or any issuing agent on behalf of the Issuer) in any manner permitted by applicable law, and in compliance with any requirements of applicable law as regards the mode of execution. A Certificate may be executed by the affixing (manually or by facsimile) of the Issuer's common seal or by the manual or facsimile signature of any Person authorised to execute it. Any Certificate bearing the facsimile signature of a Person authorised to execute it at the date of printing will be valid notwithstanding that the signatory may have ceased to be so authorised at the date of issue or at any time thereafter.

5.4 Replacement Certificates

With the consent in each case of the Holder concerned the Issuer may issue replacement Certificates in substitution for lost, stolen, damaged, defaced, destroyed or incorrect Certificates, balance Certificates where the indebtedness liabilities or obligation evidenced by a Certificate (whether a Global Certificate or otherwise) have been discharged in part, and Certificates for Registered Stock in substitution for Bearer Stock and vice versa, subject in each case to satisfactory indemnification and security therefor and to the relative terms and conditions of the Stock concerned.

5.5 **Confirmation**

At the request of a Retail Holder, or otherwise as required by the FMC Act or any other applicable law, the Issuer shall procure the Registrar on behalf of the Issuer to issue a confirmation which complies with the FMC Act and any other applicable law and is in the form agreed between the Issuer and the Registrar, or, in respect of any Stock which is quoted on the NZX Debt Market, a statement complying with the NZX listing rules (if applicable). A confirmation or statement issued in respect of Stock will not constitute a document of title.

6 **REGISTERS**

6.1 **Establishment and Maintenance**

Each Issuer shall establish and maintain in respect of the Stock issued by it a register or registers complying with all applicable laws, in which shall be entered the details required by any applicable law and such other details as the Issuer thinks fit or are required by Clause 5.2. Each Register will be kept in a form and manner enabling it to be conveniently and properly audited.

6.2 Registrar

Each Register shall be kept by a Registrar, which may be the Issuer itself (maintaining such Register among its general accounting records or separately) or any other Person (including any other Consolidated Group Member) appointed by the Issuer as Registrar in respect of the relevant Stock. Separate Registrars may be appointed in respect of separate classes or issues of Stock.

6.3 Change of Address

Each Holder (whether joint or several) other than Holders of Bearer Stock shall notify the Registrar forthwith of any change of its name or address and thereupon the relevant Register shall be altered accordingly.

6.4 Inspection

The Issuer shall ensure that, if a Retail Holder so requests, the Registrar of the relevant Stock makes available for inspection, and provides copies of or extracts from, the Register which relates to the Stock registered in the name of the Retail Holder. The Issuer and the Supervisor and any other person authorised to do so or, in this regard, contemplated by the FMC Act may inspect the Register and take copies (including electronic copies) or extracts of the same at all reasonable times during the office hours of the Registrar and subject to all applicable laws.

6.5 Closure

The Issuer may from time to time close the Register for any period or periods not exceeding in the aggregate thirty (30) days in any one year (or, if the relevant Stock is quoted on any stock exchange, a period not exceeding the times permitted by that stock exchange for listed debt securities), provided that the right of the Supervisor to inspect and take copies of the Register shall not be prejudiced by such closure.

6.6 **Audit**

Each Register shall be kept to the reasonable satisfaction of the Supervisor and audited in accordance with applicable auditing and assurance standards by the Auditors (or such other auditors as Spark NZ has appointed after consultation with the Supervisor) in the ordinary course of any periodic audit of the Consolidated Group annually within 4 months of the Issuer's balance date, and at any other time upon request by the Supervisor which must be supported by a Materiality Certificate.

6.7 New Zealand Statutory Requirements

The Issuer shall:

- (a) comply with the provisions of Sections 215 to 217 of the FMC Act as regards the maintenance and form of the Register, except (subject to Clause 6.7(c)) as regards Stock to which such provisions do not apply;
- (b) maintain each Register in Wellington or in some other place in New Zealand, except (subject to Clauses 6.7(a) and 6.7(c)) as regards Stock offered or issued solely or principally outside New Zealand where maintenance of the Register outside New Zealand is required by law or market convention;
- (c) in respect of Stock to which Sections 215 to 217 of the FMC Act do not apply or the Register for which is kept outside New Zealand, ensure (subject however to the requirements of any other applicable law) that the Register therefor is in a form that complies substantially with the requirements of Section 217 of the FMC Act and that in respect of any Register kept outside New Zealand a copy updated at such intervals as are reasonably practicable is kept in New Zealand or that other arrangements to the reasonable satisfaction of the Supervisor are made permitting the Supervisor access to the information contained in such Register.

7 TRANSFER OF STOCK

7.1 Transfer

Registered Stock may be transferred by the Holders in accordance with such procedures and on the production of such transfer and other documentation as is determined by the Issuer to be appropriate either generally or for particular Classes of Stock. In any case the transferor of Registered Stock shall (notwithstanding any provisions to the contrary as contemplated in Clause 7.3) be deemed to remain the Holder until the name of the transferee has been entered in the Register.

7.2 Death, Insanity or Bankruptcy of Holder

Any Person becoming entitled to any Registered Stock as a consequence of the death, insanity or bankruptcy of a Holder may, upon producing such evidence as to its right or title as the Directors shall require, be registered as the Holder or may transfer such Stock in accordance with the provisions of this Deed. The executors or administrators of a deceased Holder (not being one of several joint Holders) shall be the only Persons recognised by the Guaranteeing Group and the Supervisor as having any title to or interest in the Registered Stock so held by that Holder. In the event of the death of any Holder who held any Registered Stock jointly, the surviving joint Holder or Holders shall be the only Person or Persons recognised by the Guaranteeing any title to or interest in such Stock. The Guaranteeing Group and the Supervisor as having any title to or interest in such Stock. The Guaranteeing Group may retain money which is payable upon any Registered Stock to which this Clause 7.2 applies until a transmission or transfer of the Stock shall have been registered on such terms and conditions as it considers reasonable.

7.3 Terms of Stock Prevail

The above provisions of this Clause 7 shall (save as otherwise provided in the second sentence in Clause 7.1) be subject to any terms and conditions of the relevant Stock which restrict or prescribe procedures for the transfer thereof or which otherwise exclude or are inconsistent with such provisions.

PART III

CROSS GUARANTEES

8 **GUARANTEES**

Each Guaranteeing Group Member jointly and severally unconditionally guarantees to the Supervisor the due and punctual payment of the Stock Moneys by each of the other Guaranteeing Group Members (including, without limitation, the Issuer thereof in its capacity as such), as and when the same shall become due and payable in accordance with the terms and conditions of such Stock or under this Deed, and each Guaranteeing Group Member hereby agrees that the following provisions of this Clause 8 shall have effect and apply to each of the Guaranteeing Group Members and their guarantees hereunder (and in this Clause 8 the expression "guarantee" shall include every guarantee given under or pursuant to this Deed).

8.1 Payment

Whenever any default has been made by any Guaranteeing Group Member in the payment of all or any part of the Stock Moneys when due and payable in accordance with the terms thereof or under this Deed, the Guaranteeing Group Members will forthwith pay such money upon demand by the Supervisor.

8.2 Liability Not Affected

The liability of each Guaranteeing Group Member under this guarantee shall not be abrogated, prejudiced or affected by any of the following:

- (a) the granting of time, credit or any indulgence or other concession to any other Guaranteeing Group Member or other guarantor by the Supervisor or any Holder or by any compounding, compromise, release, abandonment, waiver, variation, relinquishment or renewal of any securities, documents of title, assets or of any of the rights of the Supervisor or any Holder against any other Guaranteeing Group Member or other guarantor or by anything done or omitted or neglected to be done by the Supervisor or any Holder in exercise of the authorities, powers and discretions vested in it by this Deed or by any other dealing, matter or thing which but for this provision might operate to abrogate, prejudice or affect this guarantee;
- (b) the liability of any Guaranteeing Group Member or other guarantor ceasing from any cause whatsoever (including the release or discharge by the Supervisor or the Holders);
- (c) any other person joining in this or giving any similar guarantee;
- (d) the liquidation or dissolution of the Issuer or any other Guaranteeing Group Member or any other guarantor;
- (e) any other Guaranteeing Group Member or other guarantor being incompetent to give this guarantee or any other guarantee or failing to become legally bound in whole or in part under any of them respectively;
- (f) any guarantee (including any guarantee given pursuant to this Deed) held or taken being void, defective or informal;
- (g) any Subsidiary failing or being incompetent to become or remain a Guaranteeing Subsidiary;
- (h) any alteration, variation or addition to this Deed provided that any modification of the provisions of this Clause 8 made in accordance with the provisions of this Deed shall take effect accordingly,

or by any other dealing, matter or thing which but for this provision might operate to abrogate, prejudice or affect its respective guarantee.

8.3 **Principal Obligation**

This guarantee shall be a principal obligation and shall be treated as in addition to and not in substitution for or collateral to any other security or right which the Supervisor or any of the Holders may have, and in particular shall be independent of any other right of the Supervisor or the Holders to the intent that the guarantee hereunder may be enforced against the Guaranteeing Group Members or any of them without first having recourse to any such securities or rights and without taking steps or proceedings against the Issuer or any other Guaranteeing Group Member or other guarantor and notwithstanding that any other right or security may be in whole or in part unenforceable by reason of any rule of law or equity and notwithstanding the loss by the Supervisor or any Holder of any other right or security and notwithstanding any laches, acts or omissions on the part of the Supervisor or any Holder.

8.4 **Continuing Guarantees**

Subject to Clause 9, this guarantee is to be a continuing guarantee and shall be irrevocable and shall remain in full force and effect until the whole of the Stock Moneys have been paid or satisfied.

8.5 **Discretion as to Enforcement**

The Supervisor may as regards all or any of the Guaranteeing Group Members determine from time to time whether it shall enforce or refrain from enforcing this guarantee and may from time to time make any arrangement or compromise with any one or more of the Issuer or the other Guaranteeing Group Members in relation to the premises which the Supervisor may think expedient in the interests of the Holders.

8.6 **Payment in Gross**

All moneys from time to time received by the Supervisor in reduction of the Stock Moneys from or on account of any Guaranteeing Group Member (including any dividends upon the liquidation of the Issuer or any other Guaranteeing Group Member) or other guarantor or from any other person or from the realisation of any security and capable of being applied by the Supervisor in reduction of the Stock Moneys shall be regarded as payments in gross, without any right on the part of any one or more of the Guaranteeing Group Members to stand in the place of the Supervisor in respect of or to claim the benefit of any moneys so received as against the Issuer or any Guaranteeing Group Member or other guarantor until the whole of the Stock Moneys have been paid or satisfied, so that in the event of any Guaranteeing Group Member or other guarantor liquidation the Supervisor shall be entitled to prove against it for the whole of the Stock Moneys owing.

8.7 Claims in Competition

In the event of the liquidation of any Guaranteeing Group Member or any other guarantor, no other Guaranteeing Group Member will without the consent of the Supervisor prove in such liquidation in competition with the Supervisor, and each Guaranteeing Group Member hereby authorises the Supervisor if it so elects to prove for all moneys which have been paid by such Guaranteeing Group Member hereunder or are otherwise owing to it and have not been repaid to it by the Issuer or any other Guaranteeing Group Member or any other guarantor and to retain and to carry to a suspense account and appropriate at the discretion of the Supervisor any amount received until the Supervisor shall have received one hundred (100) cents in the dollar in respect of the Stock Moneys. Each Guaranteeing Group Member hereby waives in favour of the Supervisor all rights whatever against the Issuer and each other Guaranteeing Group Member and any other guarantor or other person or their or its estate and assets so far as necessary to give effect to anything contained in this guarantee.

8.8 Non-Prejudice

This guarantee shall not prejudicially affect, or be prejudicially affected by, any other right or guarantee now or hereafter held by the Supervisor in respect of the Stock Moneys but such other right or guarantee shall be deemed to be collateral herewith and no Guaranteeing Group Member will as against the Supervisor in any way claim the benefit or seek the transfer of any such right or any part thereof or any rights of recourse.

8.9 Void or Voidable Payments

If any payment made to any Holder or the Supervisor by or on behalf of any Guaranteeing Group Member shall be avoided by law such payment shall be deemed not to have discharged or affected the liability of that or any other Guaranteeing Group Member therefor and in that event the Holder and the Supervisor and the Guaranteeing Group Members shall be restored to the position in which each would have been and be entitled to exercise all the rights which each would have had if such payment had not been made.

8.10 Suspension of Rights

A Guaranteeing Group Member shall, in respect of any sums paid by it under this guarantee and in respect of any other rights which may accrue howsoever to it in respect of any sum so paid, rank and be entitled to enforce the same only after all the Stock Moneys shall have been duly paid and satisfied.

8.11 Supervisor Under No Obligation to Marshal

The Supervisor shall not be under any obligation to marshal or appropriate in favour of any Guaranteeing Group Member or to exercise any right or power for the benefit of any such Guaranteeing Group Member until the Stock Moneys have been paid in full.

8.12 **Principal Debtor**

Although as between the Issuer and the other Guaranteeing Group Members the joint and several liability of the other Guaranteeing Group Members to the Supervisor may be that of sureties only, nevertheless as between those other Guaranteeing Group Members and the Supervisor the liability of each of those Guaranteeing Group Members shall be deemed to be the liability of a principal debtor and such liability shall not be affected or diminished by any of the matters hereinbefore mentioned or by any other act indulgence or omission which but for this present provision would or might have operated to release those Guaranteeing Group Members wholly or partly from their liabilities hereunder to the Supervisor. The validity or effectiveness of the above provisions of this Clause 8 or any other provisions in relation to or affecting Guaranteeing Subsidiaries and/or Guaranteeing Group Members shall not be affected by the fact that at the date of execution of this Deed Spark NZ is the only Guaranteeing Group Member. Upon any other Person becoming a Guaranteeing Subsidiary such provisions shall apply and have full force and effect as regards the guarantee given by and otherwise in respect of such Person as a Guaranteeing Subsidiary and/or a Guaranteeing Group Member.

9 JOINING OF GUARANTEEING SUBSIDIARIES

9.1 Joining of Guaranteeing Subsidiaries

Subject to Clause 9.3, Spark NZ undertakes that it will from time to time and with all reasonable expedition after any company becomes a Material Subsidiary procure that such Material Subsidiary shall become a Guaranteeing Subsidiary. Spark NZ may on its own volition (without being under any obligation to do so) procure any Subsidiary which is not a Guaranteeing Subsidiary to become a Guaranteeing Subsidiary.

9.2 Mode of Joining

A Subsidiary shall become a Guaranteeing Subsidiary by executing and delivering to the Supervisor a supplemental deed in or substantially in the form set out in the Second Schedule or in such other form as may be approved by the Supervisor, whereby such Subsidiary guarantees payment of the Stock Moneys and agrees to be bound as a Guaranteeing Subsidiary hereunder.

9.3 **Release of Guaranteeing Subsidiary etc.**

Notwithstanding Clause 9.1 upon receipt of a certificate of two Directors that:

- (a) there are sound commercial reasons why a Guaranteeing Subsidiary should cease to be a Guaranteeing Subsidiary or a Material Subsidiary should not be obliged to become a Guaranteeing Subsidiary; or
- (b) that a Guaranteeing Subsidiary has ceased to be a Material Subsidiary,

and (in any such case) that either releasing such Guaranteeing Subsidiary from its guarantee and covenants hereunder or permitting such Material Subsidiary to remain a Non-Guaranteeing Subsidiary (as the case may be) will not in their opinion have a materially adverse affect on the Holders generally in the reasonably foreseeable future the Supervisor shall (by the execution of appropriate release documentation to that effect) forthwith release such Guaranteeing Subsidiary from its guarantee and covenants as a Guaranteeing Subsidiary or Guaranteeing Group Member or permit such Material Subsidiary to remain a Non-Guaranteeing Subsidiary (as the case may be). A Guaranteeing Subsidiary which is at the relevant time an Issuer of outstanding Stock shall not be released hereunder unless arrangements satisfactory to the Supervisor (which may include the substitution of another Guaranteeing Group Member as Issuer in respect of such Stock) have been made to the satisfaction of the Supervisor.

9.4 "Material Subsidiary" Defined

A "Material Subsidiary" for the purposes of this Deed is a Subsidiary all of the shares of which are beneficially owned by Guaranteeing Group Members, but excluding:

- (a) any Subsidiary:
 - the profits of which, before tax and extraordinary items, are not greater than 20 per centum of the consolidated profits, before tax and extraordinary items, of the Consolidated Group determined by

reference to the Latest Financial Statements of the Subsidiary and the Consolidated Group; and

- (ii) the Total Tangible Assets of which represent not greater than 20 per centum of the Total Tangible Assets of the Consolidated Group; or
- (b) any Subsidiary whose principal business is restricted to the business of a bank, finance company or other financial intermediary or insurance company, or (in the case of a Subsidiary incorporated outside New Zealand) the business of raising and/or investing funds, or any subsidiary of any such company.

PART IV

COVENANTS

10 GENERAL COVENANTS

Each Guaranteeing Group Member covenants with the Supervisor that while any Stock Moneys remain outstanding under or in connection with this Deed:

10.1 Authorisations

It will obtain, make, do, fulfil and maintain in full force and effect all approvals, authorisations, consents, licences, registrations, notifications or other acts, conditions and things from time to time required under any applicable laws or regulations to enable it to enter into and perform its obligations under this Deed, to ensure that such obligations are legally binding and enforceable, and to make this Deed admissible in evidence in the Courts of New Zealand;

10.2 Maintain Accounts

It will keep proper accounting records in accordance with its obligations under the FMC Act (if applicable) and all other applicable laws or regulations;

10.3 Attendance at Meetings

It will permit the Supervisor through its agent or attorney or representative to attend any general meeting of Spark NZ and to be heard at any such meeting which it so attends on any part of the business of the meeting which concerns the Supervisor or the Holders;

10.4 Conduct of Business

It will carry on and conduct its business in a proper and efficient manner;

10.5 **Compliance with Obligations**

It will duly pay all its liabilities and comply with all other obligations lawfully binding on it in accordance with good commercial practice current at the relevant time;

10.6 Compliance with Legislative Requirements

It will (except to the extent exempted from compliance therewith) duly comply with all the requirements of the Financial Reporting Act, Companies Act and FMC Act, and any other applicable legislation of New Zealand or otherwise imposing the same or equivalent requirements on the Guaranteeing Group, with respect to the registration of this Deed and the lodgement of the Offer Documents (and amendments thereto), the filing of reports, accounts and statements, the registration of charges, the keeping of registers and the keeping open of the same for inspection, and all other like matters, and will generally comply with the requirements of those Acts;

10.7 Provide Offering Documents to Supervisor

It will not lodge an Offer Document in respect of Stock pursuant to the FMC Act without having provided the Supervisor with a draft copy of it in sufficient time for the Supervisor to consider and comment on it in terms of its obligations under the FMC Act or offer Wholesale Stock without having given the Supervisor reasonable notice of the intended offer and a draft of the information memorandum or other documents pursuant to which the Wholesale Stock is to be offered;

10.8 Auditors

It will facilitate the compliance by the Auditors with their duties under this Deed and under the FMC Act;

10.9 Notice of Breach

It will immediately after becoming aware thereof give notice to the Supervisor of the occurrence of any event or circumstance referred to in Clause 15.1.

11 RANKING OF STOCK AND NEGATIVE PLEDGE

11.1 Ranking

Each Guaranteeing Group Member covenants that under the law in force in New Zealand from time to time its indebtedness in respect of the Stock Moneys (whether as Issuer or as guarantor pursuant to Clause 8) will constitute unsecured unsubordinated indebtedness and will rank at least pari passu as to priority with all of its other unsecured unsubordinated indebtedness according to the tenor and the terms of the Stock, except for indebtedness preferred by statute, operation of law or this Deed.

11.2 Negative Pledge

Subject to Clause 11.3 each Guaranteeing Group Member covenants with the Supervisor that so long as any Stock remains outstanding it will not create or permit to exist any Charge over any of its Assets to secure any indebtedness or liabilities (direct or contingent), provided that this covenant shall not apply to, and the Guaranteeing Group Members shall be at liberty to create or permit to exist without breach hereof, any Charge:

- (a) Operation of Law etc.: arising by operation of law or statute in the ordinary course of business, or securing taxes or other governmental or regulatory levies, duties or imposts, or any charge in the nature of a contractor's, supplier's or vendor's lien, so long as (in each of the foregoing cases) the payment of the money secured thereby is not in default or the liability therefor of the Guaranteeing Group Member is being contested by appropriate proceedings; or
- (b) *Acquisition of Assets*: created over any Asset acquired, constructed or improved, for the sole purpose of financing or refinancing the cost of such

acquisition, construction or improvement, or over the land upon which such Asset is situated; or

- (c) Inherited Charges: Guaranteeing Subsidiaries: over any Assets of a corporation which becomes a Guaranteeing Subsidiary after the date of this Deed which existed, or which such Guaranteeing Subsidiary was contractually bound to create or enter into, at the date it became Guaranteeing Subsidiary and which was not created in anticipation of such Subsidiary becoming a Guaranteeing Subsidiary; or
- (d) Inherited Charges: over any Assets acquired by any Guaranteeing Group Member after the date upon which it became a Guaranteeing Group Member, which existed at the date of, and was not created in anticipation of the acquisition thereof by the Guaranteeing Group Member(s) concerned; or
- (e) *Supplementary Charges*: created in discharge of an obligation to supplement any Charge permitted under Clauses 11.2(b), 11.2(c), 11.2(d) or 11.2(g); or
- (f) *Substitution*: created in substitution for any Charge otherwise permitted hereunder; or
- (g) Project Charges: over assets:
 - (i) of or relating to a specific project;
 - (ii) comprising cash, deposits, financial instruments or other monetary assets,

where (in either such case) such Charge does not extend to other assets of the Guaranteeing Group and is created to secure (in the case of paragraph (i)) indebtedness incurred in respect of that project or (in the case of paragraph (ii)) new borrowings or indebtedness undertaken or incurred to raise such cash or to acquire such deposits, instruments or other monetary assets, and in either case the giving of such Charge is consistent with ordinary banking or business principles or practices then current and applicable in the relevant market and/or jurisdiction in relation to indebtedness of that nature; or

- (h) In Favour of Supervisor: in favour of the Supervisor in respect of the Stock Moneys, or in favour of any other Person where the benefit of such Charge is extended equally and rateably to the Supervisor in respect of the Stock Moneys; or
- (i) *Consent*: created with the prior written consent of the Supervisor (after, if the Supervisor so requires, approval by Special Resolution).

11.3 Charges Permitted

Notwithstanding Clause 11.2 and in addition to the Charges permitted by the provisions thereof, the Guaranteeing Group Member(s) may create or permit to exist any Charge of any nature over any of its or their Assets to secure any indebtedness if and to the extent that the aggregate principal amount of the indebtedness so

secured by all such Charges (other than any Charge referred to in Clauses 11.2(a), 11.2(c) or 11.2(d) or any Charge attaching only to Assets which are not included in the Total Tangible Assets of the Consolidated Group) does not exceed 5 per centum of the Total Tangible Assets of the Consolidated Group.

PART V

RESOLUTIONS AND REPORTS FOR SUPERVISOR

12 **REPORTING**

Spark NZ covenants with the Supervisor that it will from time to time during the currency of this Deed:

12.1 Quarterly Reports

Within seventy days (or such further time as the Supervisor may agree) after the end of each of its financial quarters furnish to the Supervisor a true copy of a certificate signed by not less than two of the Directors on behalf of them all in or to the effect of the form set out in Part I of the Third Schedule or such other form as the Supervisor may approve.

12.2 Reports

- (a) Within seventy days (or such further time as the Supervisor may agree) after the end of the first half year of, and the end of, each of its financial years furnish to the Supervisor a certificate signed by not less than two Directors on behalf of them all stating the matters referred to in Part II of the Third Schedule.
- (b) Each certificate provided under Clause 12.2(a) after the end of Spark NZ's half year or financial year shall be accompanied by a copy of any Auditors' report in respect of the consolidated financial statements prepared in respect of the Consolidated Group for that financial year and half year if such report is then available to the Directors provided that Spark NZ shall not be obliged by virtue of this provision to procure the auditing of its accounts more frequently than annually.

12.3 Financial Statements

Forthwith after Spark NZ has authorised their publication (but in any case not more than four months after the end of Spark NZ's financial year) furnish annually to the Supervisor copies of:

- (a) Consolidated Group: (i) copies of the consolidated financial statements prepared in respect of the Consolidated Group duly audited; and (ii) Spark NZ's annual report for the preceding financial year;
- (b) Guaranteeing Group: an unaudited pro forma statement showing the consolidated Assets of the Guaranteeing Group less the consolidated liabilities of the Guaranteeing Group, which statement must be signed by not less than two Directors on behalf of them all; and

(c) *Guaranteeing Subsidiary*: if requested by the Supervisor (who shall give a Materiality Certificate in respect of such request) the Latest Balance Sheet of any Guaranteeing Subsidiary.

12.3A Guaranteeing Group Audit on Request

The Supervisor may, if it considers it necessary or appropriate to safeguard Stockholders' interests in its discretion, after producing a Materiality Certificate, require Spark NZ to procure an audit of the Guaranteeing Group by a date agreed by Spark NZ and the Supervisor both acting reasonably and having regard to the time of year and state of finalisation of accounts at the time the request is made.

12.4 Auditor's Report

- (a) Subject to Clause 12.4(b), at the same time as delivery of the report required under Clause 12.2(a) after the end of Spark NZ's financial year (but in any case not more than four months after the end of Spark NZ's financial year), furnish to the Supervisor a report by the Auditors stating as of the end of such financial year:
 - (i) Breaches: whether or not in the performance of its duties as Auditors it has become aware of any breach of the provisions of this Deed or the terms of the Stock or any other matter which in its opinion is relevant to the exercise or performance of the powers or authorities conferred or imposed on the Supervisor by the Companies Act, the FMC Act, this Deed or by law and if so giving particulars thereof and in particular (but without prejudice to the generality of the foregoing) whether it has objected to any of the principles and practices applied in the preparation of the Latest Balance Sheet as at the end of that financial year;
 - (ii) Investigation required: whether or not its audit has disclosed any matter calling in its opinion for further investigation by the Supervisor in the interests of the Holders and if so particulars thereof;
 - (iii) *Register*: whether or not each Issuer or the Registrar has duly maintained the Register;
 - (iv) Director's Certificates: that it has perused the certificates given by the Directors pursuant to Clauses 12.1 and 12.2 since the last report by the Auditors given pursuant to Clause 12.4 and that, so far as matters that it has observed in the performance of its duties as Auditors are concerned, the statements and information contained in such certificates are fairly stated and the information contained in paragraph 1 of the certificates given pursuant to Clause 12.1 is correct;
 - (v) Material Changes: whether it has knowledge of any material change during the immediately preceding financial year in the method of valuation of any part of the Assets or liabilities of the Consolidated Group or any member thereof; and

- (vi) *Payments*: whether or not from normal audit tests it has conducted it is satisfied that all Stock Moneys have been paid or otherwise satisfied when due.
- (b) The Supervisor may accept, in substitution for the report required by Clause 12.4(a), a report from the Auditors that the Supervisor considers provides substantially the same information as that required by the report described in Clause 12.4(a).

12.5 Special Reports

At the request of the Supervisor (who shall give a Materiality Certificate in respect of such request) with all due expedition furnish (or cause to be furnished directly by the Auditors) to the Supervisor:

- (a) Balance Sheets: Balance Sheets of the Guaranteeing Group (and, if specifically so requested by the Supervisor, of any Guaranteeing Group Member or of the Consolidated Group) made up as at such date as may be required by the Supervisor (but being a date not less than 32 nor more than 65 days subsequent to the date of the request) relating to the period from the end of the immediately preceding financial year up to the date as at which the aforesaid Balance Sheets are made up together with all documents and reports that may reasonably be required by the Supervisor;
- (b) Auditors' Report: as soon as possible after delivery of Balance Sheets pursuant to Clause 12.5(a) above (including if required pursuant to this Deed of the Guaranteeing Group) a report by the Auditors setting out the information referred to in Clause 12.4 with such adaptations as may be necessary or such of that information as may be required by the Supervisor; and/or
- (c) *Guaranteeing Group Audit*: a report specifying the entities required to constitute the Guaranteeing Group.

12.6 Notices etc

Furnish to the Supervisor copies of any report, notice or circular at the same time as they shall be publicly distributed to any stock exchange or to the Holders by any Consolidated Group Member.

12.7 Presentation of Reports

The certificates, resolutions and reports of the Directors required pursuant to Clause 12 may be wholly or partly amalgamated in one or more documents or presented separately as the Directors may decide.

12.8 Guaranteeing Group Financial Ratio Requirements

Spark NZ shall be required to produce financial statements of the Guaranteeing Group in the circumstances set out in Clause 12.8(c) below.

(a) When tested in accordance with Clause 12.8(b):

- (i) the EBIT of the Guaranteeing Group must be at least 90% of the EBIT of the Consolidated Group; and
- (ii) the Total Tangible Assets of the Guaranteeing Group (excluding balances with other members of the Consolidated Group) must be at least 70% of Total Tangible Assets of the Consolidated Group.
- (b) For the purposes of this Clause 12.8, the above financial requirements are to be tested twice yearly, by reference to:
 - (i) financial information contained in unaudited financial statements for the Consolidated Group in respect of the first half of Spark NZ's financial year; and
 - (ii) otherwise, financial information contained in the audited annual financial statements for the Consolidated Group.
- (c) If there is a failure to satisfy one or other of the requirements described in Clause 12.8(a)(i) and Clause 12.8(a)(ii) above:
 - (i) Spark NZ shall have a period of 30 days from Spark NZ becoming actually aware of the same to take such steps or other action (or, if Spark NZ considers that the proposed steps or other action cannot be completed within such 30 day period and Spark NZ advises the Supervisor accordingly within such 30 day period, Spark NZ shall have such additional period as may be agreed to by the Supervisor) as will ensure that, if the relevant financial requirement were to be tested immediately after the taking of such step or action (by reference to the financial information referred to in Clause 12.8(b)), then such financial requirement would be satisfied; and
 - (ii) if at the end of the period provided for in (i) above Spark NZ has not taken appropriate remedial action, Spark NZ shall have a period of a further 60 days to deliver to the Supervisor unaudited financial statements for the Guaranteeing Group for either the most recently ended first half year or financial year (as the case may require). The delivery of such financial statements in accordance with this (ii) shall be deemed to remedy the failure to comply with Clause 12.8(c)(i).

13 APPOINTMENT OF AUDITOR

13.1 Consultation with Supervisor

For so long as any Retail Stock is outstanding, Spark NZ must:

- (a) before recommending the appointment or reappointment of a person as an Auditor:
 - (i) consult with the Supervisor on the appointment or reappointment; and

- ensure that any comments of the Supervisor concerning the proposed auditor are brought to the attention of the person or persons appointing or reappointing the Auditor;
- (b) notify the Supervisor if the Auditor resigns from appointment, or declines to accept appointment or reappointment, and must pass on to the Supervisor any explanation provided by the Auditor for resigning from appointment or declining to accept appointment or reappointment; and
- (c) not attempt to prevent a person who has resigned from appointment as the Auditor, or declined to accept appointment or reappointment as the Auditor, from offering an explanation, or disclosing to the Supervisor the reason, for resigning or declining appointment or reappointment.

13.2 Specified Engagement

For so long as any Retail Stock is outstanding, Spark NZ must, before recommending the appointment or reappointment of a person as the Auditor:

- (a) give the Supervisor an opportunity to be a party to an assurance engagement carried out by an auditor in relation to the Issuer's compliance with this Deed for the purpose of the Supervisor obtaining assurance of matters relevant to the exercise or performance of the powers or duties of the Supervisor; and
- (b) consult with the Supervisor on the nature and scope of any such engagement.

13.3 Terms of Appointment

For so long as any Retail Stock is outstanding, Spark NZ must ensure that the following terms are included in the terms of appointment of an auditor in its capacity as Auditor:

- (a) that the Auditor will, at the beginning of the audit, review, or engagement, give the Supervisor an opportunity to meet with the Auditor, without any representative of the Issuer being present, in order to allow the Supervisor an opportunity to raise any issues or concerns relevant to the exercise or performance of the powers or duties of the Supervisor;
- (b) that the Auditor will give the Supervisor an opportunity to meet with the Auditor, without any representative of the Issuer being present, to discuss matters arising in the performance of the audit, review, or engagement and to answer any questions the Supervisor may have concerning the audit, review, or engagement;
- (c) that the Auditor will acknowledge its obligations as auditor under the FMC Act; and
- (d) that the Auditor will carry out the audit of the audited financial statements referred to in Clause 12.3(a) in accordance with NZ GAAP and the audit of

any other reviews, engagements or other matters undertaken in relation to this Deed shall comply with New Zealand audit standards.

14 MISCELLANEOUS NOTIFICATION/REPORTING REQUIREMENTS

Each Guaranteeing Group Member covenants with the Supervisor to:

14.1 Requested information and reports

If requested by the Supervisor (or a person authorised by the Supervisor to exercise its powers), within the time and in the manner (which must be reasonable in the circumstances) specified by the Supervisor:

- (a) make available to the Supervisor all documents and records relating to the Guaranteeing Group Member; and
- (b) provide the Supervisor with any other reports or information required by the Supervisor (or other authorised person).

The reports or information may be about any matter relevant to the performance of the Supervisor's functions and include forward-looking reports.

14.2 Contravention or possible contravention of Issuer Obligations

If it has reasonable grounds to believe that it has contravened, may have contravened, or is likely to contravene any of its Issuer Obligations in a material respect, as soon as practicable:

- (a) report the contravention or possible contravention to the Supervisor; and
- (b) advise the Supervisor of the steps (if any) that the Guaranteeing Group
 Member has taken or intends to take in light of the contravention or possible
 contravention and the date by which the steps were taken or are to be taken.

14.3 Serious financial problems

If it becomes aware of information on the basis of which it could reasonably form the opinion that it is, or is likely to become, insolvent (in terms of the FMC Act), the Issuer must, as soon as practicable:

- (a) disclose to the Supervisor all information relevant to that matter that is in the possession or under the control of the Guaranteeing Group Member and that was obtained in the course of, or in connection with, the performance of its functions as Guaranteeing Group Member; and
- (b) advise the Supervisor of the steps (if any) that the Guaranteeing Group Member intends to take in respect of that matter and the date by which the steps are to be taken.

PART VI

DEFAULT AND ENFORCEMENT

15 EVENTS OF DEFAULT: ACCELERATION

15.1 Events of Default

Subject always to Clause 15.2 the Supervisor may in its discretion and shall if directed to do so by a Special Resolution of the Holders or in writing by Holders holding in aggregate not less than 20% of the Principal of the outstanding Stock, declare the Stock Moneys to be immediately due and payable upon the occurrence of any of the following events:

- (a) Default in Payment:
 - (i) Default is made for more than 5 Business Days after due date in the payment of the Principal of any Stock or for more than 14 days after due date in the payment of any other of the Stock Moneys, provided that, where in any such case such default is due to administrative failure in the banking system or to inadvertence, such default shall in any event have continued for not less than two Business Days following receipt of notice thereof by the Guaranteeing Group Member (which shall forthwith notify the Supervisor of such default) from which such payment was due; or
 - (ii) Default is made in payment of any other moneys due and payable under this Deed for 30 days after demand therefor in writing; or
- (b) Liquidation: An order of a Court is made or an effective resolution is passed or legislation is enacted for the liquidation or dissolution without liquidation of any Guaranteeing Group Member or any analogous proceedings are taken in respect of any Guaranteeing Group Member, or any Guaranteeing Group Member ceases or threatens in writing to cease to carry on the whole or a substantial part of its business, except (in any of the foregoing cases):
 - (i) for the purposes of a reconstruction or amalgamation the terms of which have been previously approved by the Supervisor (such approval not to be unreasonably withheld or delayed); or
 - (ii) where the relevant Guaranteeing Group Member is solvent at the relevant date and its Assets are to be distributed to another Guaranteeing Group Member; or
- (c) Receiverships and Execution: An encumbrancer takes possession or a receiver is appointed to the whole or any material part of the Assets or undertaking of any Guaranteeing Group Member, or a distress or execution in an amount exceeding \$10,000,000 (or its equivalent in any foreign currency) is levied or enforced upon or sued out against all or any material part of the Assets or undertaking of any Guaranteeing Group Member, except where the same is discharged or is contested by the Guaranteeing Group Member in good faith

by appropriate proceedings. For the purpose of this Clause "materiality" shall be determined by reference to whether such event has a material adverse effect on the Guaranteeing Group taken as a whole; or

- (d) Insolvency: Any Guaranteeing Group Member is unable to pay its debts as they fall due, commences negotiations with any one or more of its creditors with a view to the general readjustment or rescheduling of its general indebtedness or a material part thereof, or makes a general assignment for the benefit of or an arrangement or a composition with or for the benefit of its creditors;
- (e) Cross Default: Any indebtedness of any Guaranteeing Group Member of an amount exceeding \$10,000,000 (or its equivalent in any foreign currency) becomes due before its stated maturity by reason of default by any Guaranteeing Group Member, or is not repaid at maturity (or by the expiry of any applicable grace period), and (in either case) proper proceedings are being taken for its recovery or enforcement; or
- (f) Other Defaults: Any default on the part of a Guaranteeing Group Member shall occur under this Deed or under the terms of any Stock (other than a default specifically referred to above in this Clause 15.1) and such default continues:
 - except in the case of defaults referred to in paragraph (ii) below, for more than 30 days (or such longer period as the Supervisor may agree); or
 - (ii) in the case of a default in the provision of the reports required in accordance with Clauses 12.1 to 12.4 by the times therein specified, for more than 10 Business Days,

after receipt by such Guaranteeing Group Member of written notice from the Supervisor specifying the default and requiring the same to be remedied.

15.2 **Restrictions on Action**

Notwithstanding Clause 15.1 no occurrence of the nature specified therein shall constitute an Event of Default and the Stock Moneys shall not become immediately due and payable unless such occurrence is continuing unremedied, and (in the case of Clauses 15.1(c) to 15.1(f) inclusive) the Supervisor shall have certified to Spark NZ that in the Supervisor's opinion such occurrence is materially prejudicial to the Holders, and (in any case) the Supervisor shall have given notice to Spark NZ (which may be contained in the same document as the certificate referred to above) declaring such occurrence to be an Event of Default and the Stock Moneys to be due and payable.

16 SUPERVISOR'S POWERS OF ENFORCEMENT

Upon and following the Stock Moneys becoming immediately due and payable pursuant to Clause 15, the Supervisor may institute and pursue all such proceedings suits or other legal actions, execute any judgments obtained, exercise all such rights

of set-off or other rights or remedies available at law, compromise and effect compositions, and for all or any of the purposes aforesaid may execute and do all such assurances and things as the Supervisor may think fit on behalf of the Holders, to enforce payment of the Stock Moneys.

17 **DISTRIBUTION OF FUNDS**

All money received by the Supervisor from any Guaranteeing Group Member upon and following the occurrence of an Event of Default and the Supervisor declaring the Stock Moneys to be immediately due and payable shall (subject to payment or provision out of such funds of or for any debts or liabilities having priority to the Stock Moneys) be held and applied:

- 17.1 **Firstly**, subject to any direction made by any Court in respect thereof, in payment or retention of all costs, charges, expenses and liabilities incurred and payments made by or on behalf of the Supervisor under the provisions contained in this Deed and of all remuneration payable to the Supervisor with interest thereon as herein provided;
- 17.2 **Secondly**, in or towards payment to the Holders pari passu in proportion to the Stock Moneys owing to them respectively and in the respective currencies in which they are owed or to the liquidator (if any) of the Guaranteeing Group Member concerned or to such other lawful claimants as the High Court of New Zealand on the application of the Supervisor shall direct;
- 17.3 **Thirdly**, subject to the rights of any other creditors, in payment of the surplus (if any) of such money to the relevant Guaranteeing Group Member or as it shall direct.

18 MISCELLANEOUS PROVISIONS AS TO PAYMENT

18.1 Payment of Principal Before Interest

If the Supervisor shall be of the opinion that in the general interests of Holders it is expedient to do so, payment may be made on account of Principal before interest or the whole of the interest has been paid.

18.2 Supervisor's Powers to Postpone Distribution

If the amount of the funds at any time available under Clause 17 is less than ten per centum (10%) of the Stock Moneys owing, the Supervisor may at its discretion invest those funds in accordance with Clause 19 and such investments with the resulting income thereof may be accumulated until the accumulations together with any other funds for the time being under the control of the Supervisor and applicable for the purpose amount to a sum sufficient to pay ten per centum (10%) of the Stock Moneys as aforesaid and then such accumulations and funds shall be applied in the manner specified in Clause 17.

18.3 Notice of Distribution

The Supervisor shall give not less than twenty eight (28) days' notice to the Holders of the day place and time fixed for any payment to them under Clause 17 and after the day so fixed the Holders shall be entitled to interest only on the balance (if any)

of the money owing on the Stock held by them after deducting the amount (if any) payable in respect thereof on the day so fixed.

18.4 **Production of Certificates on Payment by Supervisor**

Subject to the terms and conditions of the relative Stock, upon and against payment under any of the provisions of Clause 17 the Certificates for the Stock in respect of which such payment is made shall be produced to the Supervisor who shall cause a memorandum of the amount and date of payment to be enfaced thereon but the Supervisor may in any particular case dispense with the production and enfacement of a Certificate upon such indemnity or security being given as it shall think sufficient or without any indemnity and may in relation to any Stock in respect of which more than one payment is made, whether by instalments or otherwise, retain such Certificates produced after the first payment for the purposes of making subsequent payments on the Stock and may make subsequent payments in the same manner as the first payment without liability for doing so on this basis.

PART VII

PROVISIONS AFFECTING SUPERVISOR

19 POWERS OF INVESTMENT

Any money held by the Supervisor subject to the trusts contained in this Deed may, at the Supervisor's discretion, be invested in the name of the Supervisor or its nominee in any investments permitted for the investment of trust funds as determined by the Supervisor (including, if Spark NZ so requests, in the purchase or subscription for Stock) with power to vary such investments and from time to time to deal with or dispose of all such investments or any part thereof. The income arising from the investments made by the Supervisor as aforesaid shall, unless and until the Stock Moneys have become immediately due and payable, belong to the Guaranteeing Group.

20 APPLICATIONS TO COURT

Having regard to any other powers or remedies available to it under this Deed or at law for the protection of the interests of Holders and to all other circumstances relevant to the general interests of Holders, the Supervisor may in its discretion apply to Court for an order:

- (a) under section 208 of the FMC Act, if the Supervisor is satisfied that:
 - the Guaranteeing Group is unlikely to be able to pay the Stock Moneys as and when due;
 - (ii) the Issuer is insolvent (as defined in the FMC Act) or the financial position or management of the Issuer is otherwise inadequate;
 - (iii) there is a significant risk that the interests of Holders will be materially prejudiced for any other reason; or

- (iv) the provisions of this Deed are no longer adequate to give proper protection to the interests of the Holders; or
- (b) under section 210 of the FMC Act and within 20 Business Days (or, with leave of the Court, within any longer period) after the passing of a Special Resolution of Holders, directing it not to comply with a Special Resolution of Holders,

and the Supervisor may assent to, approve of, or oppose any application to the Court under those sections made by the Financial Markets Authority or any Holder (where applicable) and the Supervisor shall, subject to Clause 22.8, be indemnified by the Guaranteeing Group against all expenses and liabilities incurred in relation or incidental to any such application or proceedings in priority to the Stock Moneys.

21 **POWERS OF SUPERVISOR**

- 21.1 The Supervisor:
 - (a) must:
 - (i) act honestly in acting as supervisor under this Deed;
 - (ii) in exercising its powers and performing its duties as supervisor, act in the best interests of the Holders;
 - (iii) exercise reasonable diligence in carrying out its functions as Supervisor;
 - (b) must do all the things it has the power to do to cause any contravention or alleged contravention of the Issuer Obligations in respect of the Stock to be remedied unless it is satisfied that the contravention will not have a material adverse effect on the Holders;
 - (c) subject to any court order made under section 210 of the FMC Act, must act in accordance with any direction given by a Special Resolution of Holders or an affected Class of Holders that is not inconsistent with any enactment, rule of law or this Deed in relation to:
 - (i) seeking a remedy to a contravention or alleged contravention of the Issuer Obligations in respect of the Stock; and
 - (ii) any other matter connected with the Supervisor's functions; and
 - (d) in exercising its powers and performing its duties as Supervisor, must exercise the care, diligence, and skill that a prudent person engaged in the business of acting as a licensed supervisor (as those terms are defined in the FMC Act) would exercise in the same circumstances.

22 SUPPLEMENTAL POWERS OF SUPERVISOR

By way of relief to the Supervisor additional to the provisions of the law relating to supervisors or trustees and to facilitate the discharge of its duties hereunder it is expressly declared that:

22.1 Reliance on Advice

The Supervisor in relation to this Deed may without liability for loss, obtain, accept and act on or decline and elect not to act on:

- the opinion or advice of or any information obtained from any barrister, solicitor, valuer, stockbroker, surveyor, auctioneer, chartered accountant or other expert whether obtained by the Guaranteeing Group or by the Supervisor or otherwise;
- (b) a certificate signed by any two Directors on behalf of the Directors as to any fact or matter prima facie within their knowledge or the knowledge of any Guaranteeing Group Member upon which the Supervisor may in the exercise of any of the trusts powers authorities and discretions and provisions hereof require to be satisfied, or that any particular dealing transaction step or thing is expedient or commercially desirable and not detrimental to the interests of the Holders or any Class of Holders, as sufficient evidence of such fact or matter or the expediency or desirability of such dealing transaction step or thing;
- (c) the statements contained in any certificate or certificates or in any report or reports given pursuant to the provisions of this Deed, as conclusive evidence of the facts stated therein.

22.2 Resolutions

The Supervisor shall not be responsible for acting or relying upon any resolution purporting to have been passed at any meeting of the Holders or any Class thereof in respect whereof a proper record has been made and which the Supervisor believes to have been properly and regularly passed even though it afterwards appears that such resolution is not binding or valid by reason of a defect in the convening of the meeting or the proceedings thereat or otherwise howsoever. The Supervisor shall not be liable for anything done, or omitted to be done in good faith in giving effect to a direction given to it by Holders.

22.3 Application of Funds

The Supervisor shall not be responsible for the money subscribed by the applicants for or subscribers of the Stock or be bound to see to the application thereof.

22.4 Custody of Documents

The Supervisor may place this Deed and any deeds and other documents of title relating to the Assets of the Guaranteeing Group to which the Supervisor is entitled to possession with a banker or company whose business includes undertaking the safe custody of documents or any solicitor and make such arrangements as it thinks fit for allowing the Guaranteeing Group or its solicitors or Auditors access to or possession of all or any of such deeds or documents whenever the Supervisor deems

it necessary or convenient so to do. The Guaranteeing Group agrees to pay all sums required to be paid in respect of such deposits of deeds and documents.

22.5 Defects in Title: Validity of Certificates

The Supervisor shall not be bound or concerned to enquire into or be liable for any defects in the title of any Guaranteeing Group Member to its Assets whether such defects or failure might have been discovered upon an examination or enquiry or remedied or not, and shall not be liable to any Guaranteeing Group Member or any Holder for accepting as valid any Certificate or Coupon subsequently found to be forged or not authentic.

22.6 Discretion as to Powers

Save as herein otherwise expressly provided the Supervisor shall as regards all trusts, powers, authorities and discretions vested in it by this Deed have absolute and uncontrolled discretion as to the exercise or non-exercise thereof and as to the commencement, prosecution, variation, discontinuance or compromise of any action proceeding or claim and provided it shall have acted with reasonable care and diligence it shall not be responsible for any loss, costs, damages, expenses or inconvenience that may result from the exercise or non-exercise thereof.

22.7 Delegation

The Supervisor whenever it thinks expedient in the interests of the Holders and, in relation to Retail Stock, where permitted to do so by the FMC Act or as permitted by, and then subject to, conditions imposed under the Financial Markets Supervisors Act 2011, may:

- (a) delegate at any time or times and to any Person or Persons any of the trusts powers or discretions vested in the Supervisor by this Deed which cannot conveniently be exercised by it or through its employees upon such terms and conditions and under such regulations (including the giving of power to sub-delegate) as the Supervisor may in the interests of the Holders think fit and the Supervisor shall not be responsible for any loss incurred by any misconduct or default on the part of any such delegate or sub-delegate;
- (b) instead of acting through its permanent employees in the carrying out of any of the trusts, powers or discretions hereby vested in it, employ an agent to transact all business and do all acts required to be done under this Deed including the receipt and payment of money;
- (c) delegate from time to time and at any time to any of its officers any of the trusts, powers, authorities and discretions vested in the Supervisor by this Deed which owing to the place in which they are to be or may require to be exercised cannot conveniently be exercised by the Supervisor.

22.8 Indemnity of Supervisor

(a) In relation to Wholesale Holders, without prejudice to the right of indemnity by law given to supervisors or trustees the Supervisor and every attorney, manager or agent or other person appointed by the Supervisor hereunder or under any deed supplemental or collateral hereto shall be entitled to be indemnified out of any funds held by it pursuant to this Deed and in priority to the Holders:

- (i) in respect of all liabilities and expenses incurred by it or him in the execution or purported execution of the trusts hereof or of any powers authorities or discretions vested in it or him pursuant to this Deed; and
- (ii) against all actions, proceedings, costs, claims and demands in respect of any matter or thing done or omitted or in any way relating to this Deed;

other than any of the foregoing which arise out of a wilful or negligent default or breach of trust. The Supervisor may retain and pay out of any money in its hands arising from the trusts of this Deed all sums necessary to effect such indemnity and also the remuneration and disbursements of the Supervisor as herein provided.

- (b) In relation to Retail Holders, without prejudice to the right of indemnity by law given to supervisors or trustees the Supervisor and every attorney, manager or agent or other person appointed by the Supervisor hereunder or under any deed supplemental or collateral hereto shall be entitled to be indemnified out of any funds held by it pursuant to this Deed and in priority to the Holders:
 - (i) in respect of all liabilities and expenses incurred by it or him in the execution or purported execution of the trusts hereof or of any powers authorities or discretions vested in it or him pursuant to this Deed; and
 - (ii) against all actions, proceedings, costs, claims and demands in respect of any matter or thing done or omitted or in any way relating to this Deed,

other than any of the foregoing which arise out of the Supervisor's failure to:

- (i) act honestly in acting as a supervisor;
- (ii) in exercising its powers and performing its duties as a supervisor, act in the best interests of the Holders;
- (iii) exercise reasonable diligence in carrying out its functions as a supervisor; or
- (iv) in exercising its powers and performing its duties as a supervisor, exercise the care, diligence and skill that a prudent person engaged in the business of acting as a licensed supervisor would exercise in the same circumstances,

and the Supervisor may retain and pay out of any money in its hands arising from the trusts of this Deed all sums necessary to effect such indemnity and

also the remuneration and disbursements of the Supervisor as herein provided.

22.9 Other Dealings

Nothing in this Deed shall be deemed to prohibit the Supervisor or its holding company or any of its subsidiaries or any of the subsidiaries of its holding company (all hereinafter in this clause where the context permits being included in the expression the Supervisor) or the directors or officers of the Supervisor from being a Holder or a holder of shares or other securities of any Guaranteeing Group Member or any associated company of the Guaranteeing Group or from acting in any representative capacity for a Holder or any such holder of shares or other securities. In particular and without prejudice to the generality of the foregoing, it is expressly declared that the Supervisor may so act on its own account or as executor, administrator, trustee, receiver, committee, guardian, attorney or agent or in any other fiduciary, vicarious or professional capacity. The acting in such capacity as aforesaid shall not be deemed a breach of any of the obligations hereby established or otherwise imposed or implied by law arising out of the fiduciary relationship between the Supervisor and the Holders. The Supervisor shall not by reason of its fiduciary capacity be in any way precluded from making any contracts or entering into any transactions with any Guaranteeing Group Member or any associated company of the Guaranteeing Group in the ordinary course of the business of the Supervisor, or from undertaking any insurance, financial or agency service for any of them or from accepting or holding the office of trustee for the holders of any debentures or debenture stock or other securities (whether secured or unsecured) issued by any Guaranteeing Group Member or by any other company and the Supervisor shall not be accountable either to the Guaranteeing Group Members or any other company or the Holders for any profits arising from any such contracts transactions or offices.

22.10 Represent Holders

The Supervisor may at any time or times, either of the Supervisor's own volition, or pursuant to any directions or in accordance with any policy given or indicated by any Holders or Class thereof, represent the Holders generally in any investigation, negotiation, action, transaction or proceedings affecting the interests of the Holders or that Class thereof.

22.11 Power to engage experts

- (a) The Supervisor may engage an expert (for example but not limited to, an auditor, investigating accountant, valuer, or actuary) if the Supervisor considers, on reasonable grounds, that it requires the assistance of the expert to:
 - (i) determine the financial position of the Issuer;
 - (ii) review the business, operation, or management systems, or the governance, of the Issuer; or
 - (iii) perform its functions under this Deed and/or its supervisory functions under the FMC Act or other relevant legislation.

- (b) If the Supervisor engages an expert under this Clause 22.11:
 - (i) the Issuer must provide reasonable assistance to the expert to allow the expert to provide the assistance; and
 - (ii) the fees and expenses of the expert, which must be reasonable in the circumstances, must be paid by the Issuer.

23 CHANGE OF SUPERVISOR

23.1 Retirement or removal of Supervisor

(a) **Retirement by Supervisor**

Subject to any limitation imposed by applicable statute (and in particular Section 122 of the FMC Act) the Supervisor may retire at any time without assigning any reason therefor upon giving ninety (90) days' notice in writing to Spark NZ of its intention so to do.

(b) Removal by Issuer

The Issuer may at any time without assigning any reason therefor upon giving at least ninety (90) days' notice in writing to the Supervisor (or such lesser period of notice as the Supervisor may agree), remove the Supervisor (including any new supervisor appointed upon the retirement or removal of any previous Supervisor) subject to the Financial Market Authority's consent.

(c) Removal under legislation

The Supervisor may be removed by the Financial Markets Authority or the Issuer under Part 2 of the Financial Markets Supervisors Act 2011.

(d) Special Resolution

The Holders may remove the Supervisor from office upon the passing of a Special Resolution to that effect.

23.2 Requirements for Retirement and Removal

The Supervisor may not be removed or resign under Clauses 23.1(a), (b) or (d) unless:

- (a) all functions and duties of the position have been performed;
- (b) another licensed supervisor has been appointed, and accepted the appointment, in its place and all of the moneys and investments that are held by the Supervisor under this Deed are transferred to such new supervisor; or
- (c) the Court consents.

23.3 Appointment of Substitute Supervisor

The power of appointing a substitute supervisor in place of a supervisor which has retired or has ceased to exist or to be qualified pursuant to applicable legislation for appointment as supervisor hereunder, or which or is removed from office, shall be vested in Spark NZ (subject to obtaining the approval of the Holders by Special Resolution). In the event that Spark NZ within sixty (60) days of receiving notice of the Supervisor's intention to retire or of its ceasing to exist or be qualified for appointment as supervisor hereunder or its removal from office (as the case may be) fails to exercise the power hereby vested in it of appointing a substitute supervisor or supervisors then the Holders may by Special Resolution exercise such power to the exclusion of Spark NZ.

24 SUPERVISOR'S POWERS CUMULATIVE

The powers authorities and discretions hereby conferred upon the Supervisor shall be in addition to any powers, authorities and discretions which may from time to time be vested in supervisors by law and to any powers, authorities and discretions which may from time to time be vested in the Supervisor as a Holder.

25 SUPERVISOR'S REMUNERATION

25.1 Remuneration

Spark NZ shall pay to the Supervisor by way of remuneration for its services hereunder:

- (a) *Fee*: a periodic fee of such amount and at such times as may be agreed in writing from time to time; and
- (b) *Additional*: such additional fees for convening and attending meetings of Holders as shall be agreed,

provided that:

- (c) the Supervisor may elect payment of remuneration determined by and commensurate with work actually required to be carried out by the Supervisor hereunder in lieu of the above fees;
- (d) any failure to agree on any amounts payable to the Supervisor under Clauses 25.1(a) to 25.1(c) shall be referred to arbitration in accordance with the Arbitration Act 1996 (excluding clauses 4 and 5 of the Second Schedule to that Act).

25.2 Expenses

Spark NZ will also pay all expenses (including travelling expenses) reasonably incurred by or on behalf of the Supervisor in connection with the preparation and execution of this Deed and of any deed collateral or supplemental hereto and the exercise of any power or execution of any trust conferred on the Supervisor hereunder (including without limitation the taking of any expert advice reasonably deemed necessary by the Supervisor).

25.3 Payable on Demand

All expenses incurred or payments made in the lawful exercise of the powers hereby conferred on the Supervisor shall be payable on demand.

25.4 Consents etc.

Spark NZ will also pay on demand any reasonable charges made by the Supervisor for work done by it in connection with any application under the trusts or provisions hereof for its consent to or approval of any thing act or matter.

PART VIII

MODIFICATION OF TRUST DEED AND SUBSTITUTION

26 **MODIFICATIONS**

26.1 Amendments affecting Retail Stock

(a) **Right to amend**

In the case of an amendment affecting Retail Stock, the provisions of this Deed may not be amended or replaced unless the amendment or replacement is made:

- (i) with the consent of the Supervisor; or
- (ii) (despite anything to the contrary in this Deed or in any enactment, rule of law, or agreement, including anything relating to the consent of any person to the making of amendments to this Deed) under section 109 of the FMC Act, section 22(7) or 37(6) of the Financial Markets Supervisors Act 2011 or any other power to amend or replace this Deed under an enactment.

(b) Supervisor consent

The Supervisor must not consent to an amendment to, or a replacement of, this Deed unless:

- (i) either:
 - (A) the amendment or replacement is approved by, or is contingent on approval by, the Retail Holders; or
 - (B) Spark NZ and the Supervisor are satisfied that the amendment or replacement does not have a material adverse effect on the Retail Holders; and
- the Supervisor certifies to that effect and certifies, or obtains a certificate from a lawyer, that this Deed, as amended or replaced, will comply with sections 104 to 106 of the FMC Act on the basis set out in the certificate.

(c) Holder consent

The approval of the Retail Holders for the purposes of Clause 26.1(b)(i) must be the approval of a Special Resolution of:

(i) the Retail Holders; or

(ii) each Class of Retail Holders of the Stock that is or may be adversely affected by the amendment or replacement.

26.2 Amendments affecting Wholesale Stock

In the case of an amendment affecting Wholesale Stock or any Class thereof, the Supervisor may concur with Spark NZ in making any modifications or additions to this Deed or to the terms and conditions of the relevant Wholesale Stock:

(a) No consent

Without the consent of the relevant Wholesale Holders if the same:

- (i) *Technicalities*: shall be made to correct a manifest error or is of a formal or technical nature; or
- (ii) Other Obligations: is necessary to ensure that the tenor or nature of the obligations imposed upon any Guaranteeing Group Member hereunder correspond with or do not conflict with the obligations imposed on that Guaranteeing Group Member under or pursuant to the FMC Act, the Financial Reporting Act or the Companies Act or by any other applicable statutory provision; or
- (iii) *Quotations*: shall be convenient for the purpose of obtaining or maintaining a quotation of any Stock on any stock exchange in New Zealand or elsewhere; or
- (iv) *Applicable Law*: shall be made to comply with the requirements or any modification of the requirements of any applicable law; or
- (v) *Reporting*: is in respect of any of the provisions for reporting to the Supervisor under this Deed or to Clauses 22 and 25; or
- (vi) Otherwise Appropriate: is in the Supervisor's reasonable opinion an appropriate and reasonable modification in all the circumstances in so far as they relate to this Deed and the Guaranteeing Group (as to which modifications which are necessary or convenient for complying with any practice of the New Zealand financial markets or for complying with banking practice, or are necessary convenient or useful for the purpose of any financial market, banking or business practice or technique or method adopted by either the Guaranteeing Group or Persons which carry on similar business to the Guaranteeing Group, shall be prima facie deemed reasonable and appropriate), provided that any such modification or addition is in the Supervisor's reasonable opinion not likely to have a materially adverse affect on the ability of the Guaranteeing Group to pay its liabilities due in the reasonably foreseeable future or to be or become materially prejudicial to the general interests of Wholesale Holders; or

(b) Special Resolution

If the same shall be approved by a Special Resolution of Wholesale Holders (or the relevant Class thereof).

(c) All Wholesale Holders

If the same shall be approved in writing by all the Wholesale Holders (and such authorisation may be given in one or more documents in similar form).

27 WAIVER

27.1 Temporary Variation

Subject to Clause 26 (if applicable) and any applicable law, the Supervisor may temporarily vary the provisions of this Deed or the terms and conditions of any Stock in each case for such period and on such terms as the Supervisor may deem appropriate provided that it shall be satisfied that the interests of the Holders generally will not be materially prejudiced thereby.

27.2 Waivers

Subject to any applicable law and Clause 26 (if applicable), by notice to Spark NZ the Supervisor may waive any breach or anticipated breach by any Guaranteeing Group Member of this Deed or of the terms and conditions of any Stock either wholly or in part for a specified period or indefinitely and on such other terms and conditions as it deems expedient provided that it shall be satisfied that the interests of the Holders generally will not be materially prejudiced thereby and provided further that no such waiver shall prejudice the rights of the Supervisor or the Holders in respect of any other breach.

28 CONSTRUCTION BY REFERENCE TO ANALOGOUS OBLIGATIONS AND EXEMPTIONS

Subject to Clause 26 (if applicable) and any applicable law, the Supervisor acknowledges and the parties hereto declare that any obligation of any Guaranteeing Group Member hereunder that is materially the same as or analogous to any obligation imposed upon that Guaranteeing Group Member by or pursuant to the FMC Act shall be construed subject to any relevant exemption granted or applicable to that Guaranteeing Group Member pursuant to the Act and to any modifications hereafter made to the FMC Act and shall only be enforceable against the Guaranteeing Group Member to such extent.

29 SUBSTITUTED OBLIGOR

29.1 Substitution

Subject to Clause 26 (if applicable), the Supervisor may agree, without the consent of the Holders, to the substitution in place of the Issuer of all or any Stock (*Relevant Stock*) (or of any previous Substituted Obligor under this Clause 29), as the principal debtor under this Deed in respect of the Relevant Stock, of any other body corporate incorporated in any country in the world (in this Clause 29 called the *Substituted Obligor*), provided that:

(a) Trust Deed: a trust deed is executed or some other form of undertaking is given by the Substituted Obligor to the Supervisor, in form and manner satisfactory to the Supervisor, agreeing to be bound by the terms of this Deed and the Relevant Stock with any consequential amendments that may be appropriate as fully as if the Substituted Obligor had been named in this Deed and on the Relevant Stock as the principal debtor in place of the Issuer (or of any such previous Substituted Obligor);

- (b) Maintain rights: arrangements are made satisfactory to the Supervisor for the Holders of the Relevant Stock to have or be able to have the same rights mutatis mutandis against the Substituted Obligor as they have against the Issuer (or any such previous Substituted Obligor);
- (c) *Guarantees*: the Relevant Stock continues to carry the unconditional and irrevocable guarantee of the Guaranteeing Group Members (other than the Issuer) and in addition is unconditionally and irrevocably guaranteed by the Issuer on the same terms and conditions;
- (d) Other requirements: The Issuer (or any such previous Substituted Obligor) and the Substituted Obligor comply with such other reasonable requirements as the Supervisor may direct in the interests of the Holders of the Relevant Stock, which may include a requirement that the Guaranteeing Group Members or any of them remain bound by all or certain of the provisions of this Deed in respect of the Relevant Stock in their capacities as guarantors thereof; and
- (e) *Approvals*: the Supervisor is satisfied that:
 - the Substituted Obligor has obtained all necessary governmental and regulatory approvals and consents necessary for its assumption of the obligations and liability as principal debtor under this Deed in respect of the Relevant Stock in place of the Issuer (or of any such previous Substituted Obligor);
 - (ii) where applicable the Guaranteeing Group Members and the Issuer have obtained all necessary governmental and regulatory approvals and consents necessary for the effectiveness of its or their guarantees referred to in Clause 29.1(c); and
 - (iii) such approvals and consents are at the time of substitution in full force and effect;

29.2 Release of Obligations

Any such agreement by the Supervisor pursuant to Clause 29.1 shall to the extent so expressed operate to release the Issuer or previous Substituted Obligor (as the case may be) from any or all of its obligations under this Deed in respect of the Relevant Stock. Not later than 14 days after the execution of any such documents as aforesaid and after compliance with the Supervisor's said requirements, the Supervisor shall give notice thereof to the Holders in the manner provided in Clause 31.

29.3 Amendments

Upon the execution of such documents and compliance with the said requirements the Substituted Obligor shall be deemed to be named in this Deed and on the Relevant Stock and any Coupons relating thereto as the principal debtor in place of the Issuer or any previous Substituted Obligor, and this Deed the Relevant Stock and the Coupons relating thereto shall thereupon be deemed to be amended in such manner as shall be necessary to give effect to the substitution and without prejudice to the generality of the foregoing any references in this Deed, in the Relevant Stock or in the Coupons relating thereto to the Issuer or any previous Substituted Obligor shall be deemed to be references to the Substituted Obligor.

29.4 Solvency

If any two directors or equivalent officers of the Substituted Obligor shall certify to the Supervisor that the Substituted Obligor is solvent at the time at which the said substitution is proposed to be effected the Supervisor shall not be bound to have regard to the financial condition profits or prospects of the Substituted Obligor or to compare the same with those of the Issuer or any previous Substituted Obligor.

29.5 Interests of Holders

In connection with any substitution in accordance with this Clause 29 the Supervisor shall not have regard and shall be released from any obligation or duty it might otherwise have under this Deed or implied at law in relation to the exercise of any of its discretions or powers under this Clause 29 or to advising Holders in connection with any meeting of same, to have regard to the consequences of such substitution for individual Holders or Couponholders resulting from their being for any purpose domiciled or resident in or otherwise connected with or subject to the jurisdiction of any particular territory and the Supervisor shall not be entitled to require nor shall any Holder or Couponholder be entitled to claim from the Issuer or any previous Substituted Obligor or from any other Guaranteeing Group Member any indemnification or payment in respect of any tax consequences of any such substitution upon individual Holders or Couponholders except to the extent already provided for by the terms and conditions of the Relevant Stock or the Coupons relating thereto.

29.6 Supervisor's Discretion

The exercise or non-exercise of the Supervisor's rights and powers under this Clause and the terms of any such exercise, shall be at the discretion of the Supervisor which may refuse to exercise the same without the approval of an Special Resolution of the Holders of the Relevant Stock *provided that* where the Substituted Obliger is a Guaranteeing Group Member the Supervisor shall not refuse to exercise its powers to agree to such substitution unless there is reasonable evidence indicating the likelihood that the Holders of the Relevant Stock will be materially prejudiced thereby.

PART IX

COUPONHOLDERS

30 COUPONS

30.1 General

Wherever the context permits and except as specifically provided in this Deed references to Stock include references to Coupons, references to Holders include references to Couponholders, and references to Stock Moneys include references to

moneys payable in relation to any Coupon (but not so as to treat any such amount as Principal).

30.2 Notices

Neither the Supervisor nor any Guaranteeing Group Member shall be required to give any notice to the Couponholders for any purpose under this Deed and the Couponholders shall be deemed for all purposes to have notice of the contents of any notice given to Holders as provided in Clause 31.

30.3 Assumption as to Identity of Couponholders

Wherever the Supervisor is required or entitled to exercise a duty power authority or discretion by reference to the interests of Holders pursuant to this Deed the Supervisor may assume that each Holder is the holder of all Coupons appertaining to all Stock of which it is the Holder and shall incur no liability for acting upon such assumption.

PART X

MISCELLANEOUS PROVISIONS

31 NOTICES

31.1 Notices

Any notice, demand, consent or request under this Deed shall be in writing and may be signed or given by or on behalf of any Guaranteeing Group Member, the Supervisor, or any Holder (whichever party is the giver thereof) by such Person or by an officer employee agent attorney or solicitor thereof and may be given:

- (a) To or By Holders: to Holders of Bearer Stock in accordance with Clause 31.4, and to Holders of Registered Stock by prepaid letter addressed to their relative Registered Addresses, or in the case of a notice to a corporation or to the Supervisor at its registered office or principal place of business; or
- (b) *Parties to this Deed*: by any party to this Deed to any other party (unless that other party has by 15 days written notice specified another address) by making or delivering it to that other party at the address shown below:

if to any Guaranteeing Group Member to:

Level 2, Spark City, 167 Victoria Street West, Private Bag 92028 Auckland 1142, New Zealand

Facsimile No: (09) 303 3430 Attention: General Manager Capital Markets Copied to: Facsimile No: (09) 303 3430 Attention: Company Secretary

if to the Supervisor to:

Level 2 Dimension Data House 99 Customhouse Quay PO Box 3845 Wellington 6140 New Zealand

Facsimile No: (04) 901 0108 Attention: Manager, Corporate Trusts

31.2 Effectiveness

Notices given pursuant to this Deed shall be deemed to have been made or given:

- (a) *Delivery*: in the case of delivery, when received;
- (b) *Telephone/Telex etc*: in the case of telex facsimile and other telephone or mechanical transmission, when despatched; and
- (c) Mail: in the case of posting, on the second day following the date of posting,

provided that (except in the case of notices to or by the Holders of Stock) if any notice would be deemed made or given after 4.00 p.m. on any business day in the locality of the recipient (*local business day*) or on any day that is not a local business day such notice shall be deemed made or given at 9.00am on the next succeeding local business day.

31.3 Proof of Notice

In proving the making or giving of any notice:

- (a) *Execution*: it shall not be necessary to prove that any telex facsimile or telephonically or mechanically transmitted notice was manually or originally executed by the Person making or giving it;
- (b) Addresses etc: it shall be sufficient to prove that the notice was delivered or sent properly addressed encoded numbered and stamped or (as the case shall require) in the customary manner of the method of notification adopted;
- (c) *Joint Holders*: a notice to joint Holders shall be deemed to have been properly made or served on each Holder if made or given in accordance with this Deed to the Holder whose name stands first in the Register.

31.4 Bearer Stock

Subject to any applicable terms and conditions of the Stock in question, any notice may be given by any Guaranteeing Group Member to any Holder of Bearer Stock as follows:

- (a) Conditions of issue: in the manner (if any) provided in the terms and conditions of issue of the Bearer Stock or in the absence of any such provision by advertising the same once in daily newspapers having general circulation in Auckland, Wellington, Christchurch and Dunedin, and any notice so advertised shall be deemed to be given on the day following that on which the last of such advertisements appears; and
- (b) Other Legal Ways: in any other manner permitted by law.

At the date of this Deed it is expected that notices under Clause 31.4(a) will be given in the New Zealand Herald, The Dominion Post, the Press and the Otago Daily Times respectively.

32 MEETINGS OF HOLDERS

32.1 Convened by Spark NZ

At the request in writing of the Supervisor or of any Holder(s) holding at least five (5) per centum in Principal amount of the Stock of that Class Spark NZ shall summon a meeting of the Holders or the Holders of any Class for the purpose of considering the Latest Balance Sheet of the Consolidated Group or any member thereof and of giving directions to the Supervisor in relation to the exercise of its powers under this Deed and applicable law. Spark NZ may having first notified the Supervisor of its intention to summon the meeting and the time and place thereof, summon a meeting of the Holders or any Class thereof for such purpose as Spark NZ considers appropriate.

32.2 Convened by the Supervisor

At any time the Supervisor (who shall give Spark NZ a Materiality Certificate in respect thereof) may summon a meeting of Holders or any Class of Holders to consider such business as shall be placed before such meeting.

32.3 Other Rules

Subject to compliance with any applicable laws, the Supervisor may without the consent of the Holders agree with Spark NZ to the adoption of supplemental rules or procedures for meetings of Holders and/or variations to the rules and procedures applying to such meetings set out in the Fourth Schedule. Subject to the foregoing, meetings of Holders shall be conducted in accordance with the rules and procedures set out in the Fourth Schedule.

33 RELEASE UPON PAYMENT

Upon proof being given to the reasonable satisfaction of the Supervisor that all of the Stock Moneys have been paid or satisfied or that provision for such payment or satisfaction has been made in accordance with the provisions of this Deed and upon payment or retention of all costs charges and expenses incurred by or payable to the Supervisor in relation to this Deed and the remuneration of the Supervisor and all other money payable hereunder the Supervisor shall at the request and cost of Spark NZ execute a Deed of Release of this Deed and shall thereupon retire.

IN WITNESS WHEREOF this Deed is executed by the parties on the date first written above.

FIRST SCHEDULE

Certificate No:

Nominal Amount of Stock: \$

SPARK NEW ZEALAND LIMITED [ISSUER'S NAME] LIMITED

(the *Issuer*)

incorporated in []

Registered Office:

Stock Registrar:

ACCOMMODATION STOCK CERTIFICATE

The Stock comprised in this Certificate is unsecured Accommodation Stock created and issued pursuant to the constitution of the Issuer and constituted by a trust deed dated the day of made between [the Issuer/Spark New Zealand Limited (the *Company*)] of the first part and The New Zealand Guardian Trust Company Limited (as Supervisor for the Holders) of the second part (such trust deed, as modified or supplemented from time to time being referred to herein as the *Trust Deed*) and is issued with the benefit and subject to the provisions of the said Trust Deed [and the sub-joined Conditions].

THIS IS TO CERTIFY THAT

- 1 _____ (hereinafter called *the Holder*) is the registered holder of \$ in nominal amount of Stock (hereinafter called *the said Stock*).
- 2 Subject to the provisions of Clause 3 hereof the said Stock is issued by way of continuing cover for the due payment and satisfaction to the Holder (upon due demand) of all accommodation indebtedness liabilities and obligations of the Issuer whether incurred before or after the issue of this Certificate and whether matured or not and whether incurred by the Issuer alone or jointly with another or others and whether as principal or surety and whether absolute or contingent, in respect of [define relevant indebtedness] (hereinafter called the *Total Amount*).
- 3 The said Stock is issued on the condition that although the nominal amount of the Stock is as stated in this Certificate the Holder shall be entitled to recover from the Issuer the Total Amount actually owing to the Holder at the relevant time including interest from time to time payable to the Holder pursuant to and in terms of the obligations described in Clause 2.

DATED the day of

Signed for and on behalf

of [Issuer]

by:

NOTES

- (1) All sums specified in this Certificate relate to New Zealand currency.
- (2) Attention is drawn to the Conditions endorsed hereon.

CONDITIONS

- 1 The holders of Accommodation Stock are entitled pari passu and rateably with the holders of all other Stock constituted by the said Trust Deed to the benefit of and are subject to the provisions of the said Trust Deed.
- Subject as hereinbefore provided, each holder of Accommodation Stock shall as a condition precedent to receiving payment of the amount covered by its Stock Certificate or any part thereof after the Stock Moneys have been declared immediately due and payable produce this Certificate for endorsement to the Company or to the Supervisor as the case may require PROVIDED ALWAYS that this Condition shall not preclude payment by the Issuer or by the Supervisor with the consent of the Issuer without production of this Certificate if the Issuer or the Supervisor so elects AND PROVIDED FURTHER that this Condition shall not apply if the Supervisor requires payment to it of the said amount in accordance with the provisions of the said Trust Deed.
- 3 Subject to the provisions of the said Trust Deed the Issuer will only recognise the Registered Holder of any Accommodation Stock as the absolute owner thereof and except as ordered by a Court of competent jurisdiction or by statute required shall not be bound to take notice or see to the execution of any trust whether express or implied or constructive to which the Accommodation Stock may be subject and the

receipt of the Registered Holder for the amount payable in respect thereof shall be a good discharge to the Issuer notwithstanding any notice it may have whether express or otherwise of the right title interest or claim of any other person to or in respect of the amount so payable. No notice of any trust express implied or constructive shall be entered on the Register in respect of any Accommodation Stock.

SECOND SCHEDULE

(Form of Supplemental Deed - Clause 9)

THIS SUPPLEMENTAL DEED is made this day of

BY

[] Limited a duly incorporated company having its registered office at [] (the Subsidiary)

The New Zealand Guardian Trust Company Limited a company incorporated in New Zealand and having its registered office at Auckland (*the Supervisor*)

- A The Subsidiary is a subsidiary within the meaning of the Companies Act 1993 of Spark New Zealand Limited (*Spark NZ*).
- B Spark NZ and the Supervisor have entered into a trust deed dated the 25th day of October 1988 (such trust deed, as from time to time modified supplemented or novated in accordance with its terms being referred to herein as the *Trust Deed*).
- C At the request of Spark NZ the Subsidiary wishes to execute this Deed (being a deed supplemental to the Trust Deed) in order to become a Guaranteeing Subsidiary as defined in the Trust Deed pursuant to provisions in that behalf therein contained.
- D The board of Directors of the Subsidiary, pursuant to the provisions of its Constitution has resolved that the Subsidiary guarantee the due payment by Spark NZ and each other Guaranteeing Group Member of all money from time to time owing under the Trust Deed AND that this Deed be accordingly executed as a deed in accordance with section 180(1)(a) of the Companies Act 1993.

NOW THEREFORE THIS SUPPLEMENTAL DEED WITNESSES as follows:

1 DEFINITIONS AND INTERPRETATIONS

To the extent to which the same are applicable, the definitions and provisions contained in Clause 1 of the Trust Deed shall apply to and be incorporated in this Supplemental Deed.

2 ACKNOWLEDGEMENT BY SUBSIDIARY

The Subsidiary hereby appoints the Supervisor (and the Supervisor hereby accepts appointment) on the same terms as set out in the Trust Deed and the Subsidiary acknowledges that Stock has been and will henceforth be taken up by the Holders on the condition and in part consideration that the Subsidiary will give or has given to the Supervisor the guarantee hereinafter contained by way of security for the obligations constituted by the Trust Deed or intended so to be.

3 GUARANTEE

The Subsidiary hereby unconditionally and irrevocably guarantees, jointly and severally with each other Guaranteeing Group Member, to the Supervisor the due and punctual payment of the Stock Moneys by each of the other Guaranteeing Group Members (including, without limitation, the Issuer thereof in its capacity as such) as and when the same shall become due and payable in accordance with the terms and conditions of such Stock or under the Trust Deed, and the Subsidiary hereby agrees that all of the provisions of Clause 8 of the Trust Deed shall be incorporated in this Supplemental Deed and shall have effect and shall apply to this guarantee.

4 INCORPORATION OF TERMS

Pursuant to Section 14 of the Property Law Act 2007 it is declared that there shall be deemed to be incorporated in this Supplemental Deed all the covenants, undertakings, powers, obligations and other provisions of the Trust Deed and the schedules thereto relating to or affecting the Guaranteeing Subsidiaries or Guaranteeing Group Members in the same manner and to the same extent as if the same had been mutatis mutandis set out in full in this Supplemental Deed and made applicable to the Subsidiary, and the Subsidiary accordingly covenants, jointly and severally with each other Guaranteeing Group Member, in favour of the Supervisor to duly perform and observe and be bound by the said covenants, undertakings, powers, obligations and other provisions imposed on or relating to or affecting it by or under this Deed or the Trust Deed or by the terms and conditions of any Stock.

5 GOVERNING LAW

This Supplemental Deed shall be governed by and construed in accordance with the laws of New Zealand.

IN WITNESS WHEREOF the Subsidiary has duly executed this Supplementary Deed the day and year first above written.

Executed and delivered as a Deed

Signed for and on behalf of [Subsidiary] by:

The New Zealand Guardian Trust Company Limited by

Authorised Signatory

Authorised Signatory

THIRD SCHEDULE

PART I

(Quarterly Directors' Report Clause 12.1)

TO: The New Zealand Guardian Trust Company Limited

This certificate is given to you in your capacity as the Supervisor under the Trust Deed dated 25 October 1988 (as amended and restated from time to time) entered into between Spark New Zealand Limited (*Spark NZ*) and yourself as Supervisor, pursuant to Clause 12.1 thereof. Terms defined in the Trust Deed have the same meanings herein.

- 1 The Directors of Spark NZ note and certify to the Supervisor, on behalf of all the Directors of Spark NZ, that as at the close of business on [* here insert the last day of the last preceding financial quarter] (the *reporting date*) by reference to the Latest Financial Statements as at the reporting date:
 - (a) The aggregate Principal amount of the outstanding Stock was [\$];
 - (b) The amount of the Consolidated Group's Total Tangible Assets was [\$];
 - (c) The aggregate of the principal amounts secured by Charges on the Assets of the Guaranteeing Group was:
 - (i) Charges referred to in Clauses 11.2(a), 11.2(c) and 11.2(d) of the Trust Deed [\$];
 - (ii) Other Charges [\$];

As to charges referred to in Clause 11.2(a) of the Trust Deed the above certification refers only to Charges of which the Directors who are members for the time being of the Audit and Risk Management Committee of the board of Directors of Spark NZ are actually aware; it is possible that other Charges of this nature may have been created or arisen.

- (d) The aggregate amount of the Consolidated Group's liabilities (being the amounts shown as liabilities on the Consolidated Group's consolidated balance sheet comprised in the Latest Financial Statements in accordance with NZ GAAP) was [\$].
- 2 The Directors of Spark NZ have resolved that they are satisfied that:
 - (a) The Stock that was issued in the quarter that ended on the reporting date was issued without breach of the Trust Deed; and
 - (b) Having received such reports projections certificates and assurances as they consider necessary and having regard to the financial position of the Guaranteeing Group (including contingent liabilities) as a going concern which such Directors are satisfied will be the case over the relevant period, the

anticipated trading and transactions during the next twelve months, and the sources of finance arranged or capable of being arranged during such twelve months, the Guaranteeing Group will be able to meet all its liabilities (including, maturing Stock and interest thereon) as they fall due or are anticipated to become payable during such twelve months.

This certificate is signed by two Directors on behalf of all of the Directors:

PART II

(Semi-Annual Directors' Certificate pursuant to Clause 12.2(a))

Such certificate shall detail in relation to the Guaranteeing Group (unless otherwise expressly stated), as at the close of business on the last day of the preceding financial half-year (the *reporting date*), by reference to such management or other accounts and information as the Directors consider appropriate, any matters that since the reporting date of the last of such certificates (or, in the case of the first of such certificates, since the Effective Date) have in their opinion occurred to adversely affect the interests of the Holders generally and without prejudice to the generality thereof stating to the best of their knowledge and belief after having made all due enquiries whether or not:

- (a) Payments: the Guaranteeing Group has duly observed and performed all the covenants conditions agreements and provisions binding upon it under this Deed including its Issuer Obligations (and, in particular, in relation to the Register) and the payment of all Stock Moneys on their due dates;
- (b) *Accounting*: there have been any material changes in any accounting method or method of valuation of Assets or liabilities;
- (c) *Losses*: any material trading or capital loss shall have been sustained by the Guaranteeing Group or any Member thereof and if so particulars thereof;
- (d) Material matters: any matters which since the reporting date referred to in the last of such certificates (or, in the case of the first of such certificates, since the Effective Date) have occurred that in the Directors' opinion will materially and adversely affect the interests of the Holders and if so particulars thereof;
- (e) Adverse Matters to Trust Deed: there shall have been any matter, and state if there is no such matter, which has arisen in relation to Spark NZ or the Guaranteeing Group (including each Issuer in its capacity as Issuer) which would materially and adversely affect the ability of Spark NZ or the Guaranteeing Group (including each Issuer in its capacity as Issuer) respectively to perform their respective obligations under this Deed in respect of the Retail Stock or which materially and adversely affects the Holders of Retail Stock;
- (f) Nature of business: there shall have been any material change in the nature of the business of Spark NZ or of the Consolidated Group or Guaranteeing Group (taken as a whole) and if so particulars of that change;

- (g) *Breaches*: there shall have been any material breach of the provisions of this Deed (including an Event of Default) or of the terms of any Stock;
- (h) *Future liabilities*: any circumstances are known to them that materially adversely affect the ability of the Consolidated Group or of the Guaranteeing Group to pay its liabilities due within the next six months;
- (i) Contingencies: any material contingent liabilities shall have been incurred by any Guaranteeing Group Members (other than in the ordinary course of business) and if so the amount thereof;
- (j) *New Subsidiaries*: any Subsidiaries have been formed or acquired, and if so details thereof and whether such Subsidiaries are Material Subsidiaries;
- (k) Insurance: the Guaranteeing Group continues to insure its assets on a basis which is materially the same as that last disclosed to the Supervisor and, if not, details of any material changes to such basis which have been approved by the Directors of Spark NZ, and all premiums owing in respect of such insurance have been paid;
- (I) FMC Act: in respect of any Stock offered pursuant to a regulated offer (in terms of the FMC Act) in New Zealand since the last reporting date, the Guaranteeing Group has complied with all applicable provisions of the FMC Act (including any relevant exemption granted or applicable) and, in respect of any such Stock issued and for which an Offer Document was required in terms of the FMC Act (i) such Stock has been issued on the basis of the Offer Document/s; and (ii) the Offer Document/s dated [here insert dates of the Offer Document/s current at the date of the certificate] is/are not known to the Directors to be false or misleading in any material particular by reason of failing to refer, or give proper emphasis, to adverse circumstances;

Such certificate shall also include the following statements:

(m) As at the reporting date:

Total Tangible Assets of the Consolidated Group (B) \$[]

Ratio of (A) to (B) = []% (minimum: 70%, or such higher % as may be substituted therefor by the operation of clause 12.8(a) of the Trust Deed). [*to be included if the minimum requirement is not satisfied:* Spark NZ has taken or proposes to take the following steps or action to remedy such deficiency [*insert details here*]].

(n) In respect of the [six month/twelve month] period ending on the reporting date:

EBIT of Guaranteeing Group (C)	\$[]	
EBIT of Consolidated Group (D)		\$[]

Ratio of (C) to (D) = []% (minimum: 90%, or such higher % as may be substituted therefor by the operation of clause 12.8(a) of the Trust Deed). [*to be included if the minimum requirement is not satisfied:* Spark NZ has taken or proposes to take the following steps or action to remedy such deficiency [*insert details here*]].

(Financial statements pursuant to Clause 12.3)

[In the case of the reporting date immediately following the end of its financial year, to be provided forthwith after Spark NZ has authorised their publication (but in any case not more than four months after the end of Spark NZ's financial year)] Attached are:

- a copy of the audited consolidated financial statements for the Consolidated Group and Spark NZ's annual report, in each case for the immediately preceding year, each as required to be furnished to the Supervisor under clause 12.3(a) of the Trust Deed; and
- a copy of the Auditor's report in respect of the consolidated financial statements for the Consolidated Group for the immediately preceding year, as required to be furnished to the Supervisor under clause 12.2(b) of the Trust Deed; and
- (iii) a copy of the Auditor's report required to be furnished to the Supervisor under, and stating the various matters specified in, clause 12.4 of the Trust Deed; and
- (iv) a copy of the unaudited pro forma statement showing the consolidated Assets of the Guaranteeing Group less the consolidated liabilities of the Guaranteeing Group required to be furnished to the Supervisor under clause 12.3(b) of the Trust Deed.

FOURTH SCHEDULE

MEETINGS OF HOLDERS

1 CONVENING

1.1 References in this Schedule to *Holders* shall include only the Holders comprising the Class or Classes for which any meeting is convened except where the context indicates otherwise and "Stock" shall be construed accordingly. References in this Schedule to "regulations" are references to the provisions of this Schedule.

Appointed Time means the day and time at which any meeting of Holders or the taking of a poll of Holders (not at a meeting of Holders) is due to be held.

Proxy Closing Time means 48 hours before the Appointed Time of the relevant meeting of Holders or taking of a poll of Holders.

- 1.2 The Supervisor or Spark NZ or any other Issuer may at any time convene a meeting of all or any Class of Holders. Spark NZ or any other Issuer shall whenever required to do so pursuant to clause 32 of the Trust Deed or by the FMC Act or FMC Regulations convene a meeting of the Holders.
- 1.3 Notwithstanding the provisions of regulation 1.2 above the Supervisor shall not be obliged to convene a meeting of Holders pursuant to such provisions until it has been indemnified to its satisfaction against all costs and expenses to be thereby incurred.
- 1.4 Before Spark NZ or any other Issuer convenes a meeting it shall advise the Supervisor in writing of the place day and hour of the meeting and the nature of the business to be transacted, and it shall obtain the prior approval of the Supervisor to the draft of the documents to be sent to the Holders in respect of such meeting, and if the Supervisor shall so require Spark NZ or the other Issuer shall include with the documents sent any statement which the Supervisor requires to make in relation to the meeting and the matters to be dealt with at such meeting.

2 **PLACE**

Meetings shall be held in the city or town at which the registered office of Spark NZ is situated or at such other place as the Supervisor approves.

3 NOTICE OF MEETING

- 3.1 Spark NZ must ensure that written notice of the time and place of a meeting of Holders is sent to the following at least 15 Business Days before the meeting:
 - (i) every Holder entitled to receive notice of the meeting;
 - (ii) the Supervisor; and

- (iii) every Director and an Auditor of the Issuer.
- 3.2 The notice must state:
 - (i) the nature of the business to be transacted at the meeting in sufficient detail to enable a Holder to form a reasoned judgment in relation to it;
 - (ii) the text of any Special Resolution to be submitted to the meeting; and
 - (iii) the right of a Holder to appoint a proxy.
- 3.3 If a Special Resolution is to be submitted to the meeting:
 - a draft of the proposed notice of the meeting must be given to the Supervisor at least 10 Business Days before the notice is given under regulation 3.1 (or any lesser period approved by the Supervisor); and
 - (ii) the notice of the meeting must be accompanied by a document containing the Supervisor's comments on the proposed Special Resolution (but only if the Supervisor has provided those comments in writing to the Issuer at least 5 Business Days before the notice is given under regulation 3.1).
- 3.4 An irregularity in a notice of a meeting is waived if:
 - all Holders entitled to attend and vote at the meeting attend the meeting without protest as to the irregularity, or if all such Holders agree to the waiver; or
 - (ii) the Supervisor indicates at the meeting that the Supervisor is satisfied that the irregularity has not resulted in and is unlikely to result in any material prejudice to the Holders.
- 3.5 The accidental omission to give notice of a meeting to, or the failure to receive notice of a meeting by, a Holder does not invalidate the proceedings at that meeting.
- 3.6 If a meeting of Holders is adjourned for less than 30 days, it is not necessary to give notice of the time and place of the adjourned meeting other than by announcement at the meeting that is adjourned.

4 QUORUM

- 4.1 No business may be transacted at a meeting of Holders if a quorum is not present.
- 4.2 A quorum for a meeting of Holders at which a Special Resolution is to be submitted is present if Holders or their proxies are present who hold Stock with a combined Principal of no less than 25% of the Principal of the Stock held by those persons who are entitled to vote on the business to be transacted by the meeting.

- 4.3 A quorum for any other business at a meeting of Holders is present if Holders holding at least 10% of the Principal of the Stock are present in person or by proxy and in any case at least 2 Holders or their proxies must be present.
- 4.4 Despite subclause 4.1, if a quorum is not present within 30 minutes after the time appointed for the meeting:
 - (i) in the case of a meeting called under section 120(1)(b) or (c) or 161(1)(b) or (c) of the FMC Act, the meeting is dissolved; and
 - (ii) in the case of any other meeting, the meeting is adjourned to the day that is 10 Business Days after the date appointed for the meeting at the same time and place, or to such other date, time, and place as the Supervisor may appoint, and if, at the adjourned meeting, a quorum is not present within 30 minutes after the time appointed for the meeting, the Holders or their proxies present are a quorum.
- 4.5 To avoid doubt, a Holder participating in a meeting by means of audio, audio and visual, or electronic communication is present at the meeting and part of the quorum.

5 **RIGHT TO ATTEND AND SPEAK**

Any director, officer or solicitor of the Supervisor and any other person authorised in that behalf by the Supervisor and any Director or the Secretary or Auditor or solicitor of Spark NZ or the other Issuer or any other person authorised in that behalf by Spark NZ or the other Issuer may attend any meeting and all such persons shall have the right to speak at the meeting.

6 CHAIRMAN

A person nominated in writing by the Supervisor shall preside as chairman at every meeting and if no such person is nominated or if at any meeting the person nominated is not present within fifteen minutes after the time appointed for holding the meeting the Holders present and eligible to vote thereat shall choose one of their number to be chairman and failing such choice Spark NZ or the other Issuer may appoint the chairman.

7 ADJOURNMENT

- 7.1 The Chairman may with the consent of any meeting at which a quorum is present and shall if so directed by the meeting adjourn the meeting from time to time and from place to place.
- 7.2 No business shall be transacted at any adjourned meeting except business which might have been lawfully transacted at the meeting from which the adjournment took place.

8 PERSON ON REGISTER

In the case of Registered Stock the persons registered as Holders in the Register at Proxy Closing Time and no other person or persons shall be recognised and treated as the legal holders of the Stock specified in the Register whether such persons are or are not in fact the beneficial owners of such Stock.

9 **AUTHORITY TO VOTE**

- 9.1 The following persons shall except as may be otherwise provided by regulation 13.8 of this Fourth Schedule be exclusively entitled to vote in person or by Representative in respect of the Stock or Class thereof in respect of which the meeting is convened:
 - (a) in the case of Registered Stock the persons registered as Holders in the Register at Proxy Closing Time or the personal representative or assignee in bankruptcy of any such Holder, or their respective Representatives;
 - (b) in the case of Bearer Stock, the bearers of the relevant Certificates therefor or their respective Representatives.
- 9.2 For the purposes of this Fourth Schedule a Representative need not be a Holder and shall be entitled to attend meetings of Holders, be counted in a quorum, vote, join in any demand for a poll and exercise all such other powers in relation to the Stock, in respect of which he or she has been so appointed or nominated as though he or she were in relation thereto the Holder present in person.
- 9.3 *Closure of Register*: For the purpose of establishing voting entitlements at a meeting the Register shall be closed as of the close of business on the business day immediately preceding the day on which the Proxy Closing Time falls and shall remain closed until after the relevant meeting has been closed or adjourned.
- 9.4 *Representative*: For the purpose of this Fourth Schedule, a "Representative" shall be:
 - (a) in the case of Registered Stock, a proxy or an attorney or a corporate representative appointed in the manner referred to in regulations 10, 11 or 12 as the case may be;
 - (b) in the case of Bearer Stock, a person appointed in the manner set out in regulation 9.5 or in such other manner as may be prescribed in the terms and conditions of such Stock and endorsed on the relevant Certificate or as may be approved by the Supervisor and notified to the Holders in the notice of meeting given pursuant to regulation 3.1.

The Supervisor or any officer of the Supervisor may be appointed a representative of a Holder.

- 9.5 *Voting Authorities for Bearer Stock*: Subject to the relevant terms and conditions of the Stock, Holders of Bearer Stock may appoint Representatives in the following manner:
 - Holders of Bearer Stock, upon presentation of their Bearer Stock Certificate to any registered bank in New Zealand or the Supervisor (at any of their respective branches), may request such bank or the Supervisor, as the case may be, to issue a voting authority (*a Voting Authority*) in respect thereof; and
 - (ii) such bank or the Supervisor may issue to such Holder a Voting Authority in the following form, or in such other form as the Supervisor shall determine, namely:
 - TO: SPARK NEW ZEALAND LIMITED VOTING AUTHORITY for the Meeting of Holders of Stock to be held on or any adjournment thereof.

Serial Number(s) of Stock Certificate(s):

Nominal amount of Bearer Stock:

The Undersigned certifies that the above Bearer Stock Certificate(s) have been deposited with the undersigned and will not be released except upon surrender of this Voting Authority in exchange for the relevant Certificate(s).

SIGNED on behalf of:

Authorised Signatory

Voting Authorities shall be valid only for the specific Meeting to which they refer or any adjournment thereof.

10 **PROXIES**

- 10.1 The holders of Registered Stock may appoint proxies.
- 10.2 An instrument of proxy shall be in the following form or a form as near thereto as circumstances admit or in such other form as the Supervisor shall approve and need not be witnessed:

I/We, OF

amount of Stock of Spark New Zealand Limited hereby appoint

..... of of or failing him or failing him the Chairman of the meeting as my/our proxy to vote for me/us and on my/our behalf at the meeting of Holders of Stock of Spark NZ to be held on , and at any adjournment thereof.

I/We authorise my/our proxy to vote

*in favour of/against the resolution(s).

Signed this day of .

Signature of Holder(s).....

*Strike out whichever is not desired. If neither "in favour of" nor "against" is struck out, the proxy will vote as he thinks fit.

- 10.3 The instrument appointing a proxy shall be in writing under the hand of the appointor or of his attorney duly authorised in writing or if the appointor is a company signed by an officer or attorney so authorised or by any director, secretary, general manager, investment manager or other person who appears to the Supervisor to have authority to sign the instrument.
- 10.4 A person appointed to act as a proxy need not be a Holder and a holder of a proxy shall have the right to speak at the meeting.
- 10.5 The instrument appointing a proxy and the power of attorney or other authority (if any) under which it is signed or a copy of such power or authority certified by [a Notary Public] or in such other manner as the Supervisor shall approve shall be deposited at such place as the Supervisor, or Spark NZ or the other Issuer with the approval of the Supervisor, may in the notice convening the meeting direct (or if no such place is appointed then at the registered office of Spark NZ) not later than Proxy Closing Time and in default the instrument of proxy shall not be treated as valid provided that the Supervisor may in its discretion accept and treat as valid an instrument of proxy delivered late or in such other form satisfactory to the Supervisor.
- 10.6 A proxy shall unless the contrary is stated thereon be valid as well for any adjournment of the meeting as for the meeting to which it relates. Notwithstanding any provision contained in an instrument of proxy no instrument of proxy shall be valid after the expiration of twelve months from the date of its execution but this provision shall not be construed to apply to the appointment of an attorney or representative otherwise than by an instrument of proxy.
- 10.7 An instrument of proxy in favour of the chairman of the meeting (howsoever expressed) shall be valid and effectual as though it were in favour of a named person and shall constitute the person who chairs the meeting for which the proxy is used (whether an adjournment of not) the lawful proxy of the appointer.

- 10.8 A person appointed proxy shall have the right to demand or join in demanding a poll and shall (except to the extent to which the proxy is specially directed to vote for or against any proposal) have the power generally to act at the meeting for the Holder concerned.
- 10.9 A vote give in accordance with the terms of an instrument of proxy or power of attorney or authority as referred to in regulation 12.1 (or other form approved by the Supervisor) shall be valid notwithstanding the previous death or insanity or liquidation of the principal, or revocation of the proxy or power of attorney or authority (or of the authority under which the proxy power of attorney or authority was executed), or the transfer of the Stock in respect of which the vote is given, provided that no intimation in writing of such death insanity liquidation revocation or transfer is received by Spark NZ at its registered office before the commencement of the meeting or adjourned meeting at which the proxy power of attorney or authority is used.

11 ATTORNEYS

Any Holder may by power of attorney appoint an attorney (who need not be a Holder) to vote and act on his behalf at any meeting. An attorney shall be entitled to produce evidence of his appointment at any time before the Appointed Time of the meeting or adjourned meeting or for the taking of a poll at which he proposes to vote, or at such meeting or adjourned meeting. Such Attorney as so empowered may appoint a proxy for the Holder granting the power of attorney.

12 CORPORATE REPRESENTATIVES

A person authorised by the Holder being a company to act for it as its representative at any meeting or any adjourned meeting shall be entitled to exercise the same powers on behalf of that company as that company could exercise if it were an individual Holder. A person so authorised shall be entitled to produce evidence of his appointment or authority at any time before the Appointed Time of the meeting or adjourned meeting or for the taking of a poll at which he proposes to vote, or at such meeting or adjourned meeting.

13 **VOTING**

13.1 A resolution put to the vote of a meeting shall be decided on a show of hands unless a poll is (before or on the declaration of the result of the show of hands) demanded by the chairman or the Supervisor or any representative of the Supervisor or by one or more Holders holding or representing not less than 5 per centum in nominal amount of the Stock for the time being outstanding or (as the case may be) of the appropriate Class of Stock. Unless a poll is so demanded a declaration by the chairman that a resolution has been carried or carried unanimously or by a particular majority or lost shall be conclusive evidence of the fact without proof of the number or proportion of the votes recorded in favour of or against such resolution.

- 13.2 If a poll is duly demanded it shall be taken in such manner as the chairman may direct and the result of such poll shall be deemed to be the resolution of the meeting at which the poll was demanded.
- 13.3 In case of an equality of votes whether on a show of hands or on a poll the chairman of the meeting at which the show of hands takes place or at which the poll is demanded shall be entitled to a casting vote in addition to the votes (if any) to which he may be entitled as a Holder or on behalf of Holders.
- 13.4 A poll demanded on the election of a chairman or on a question of adjournment shall be taken forthwith. A poll demanded on any other question shall be taken either immediately or at such time (not being more than thirty days from the date of the meeting) and place as the chairman may direct. The result of such poll shall be deemed to be the resolution of the meeting at which the poll was demanded. No notice need be given of a poll not taken immediately.
- 13.5 The demand for a poll shall not prevent the continuance of a meeting for the transaction of any business other than the question on which the poll has been demanded.
- 13.6 On a poll votes may be given either personally or by Representative. On a poll a person entitled to more than one vote need not use all his votes or cast all the votes he uses in the same way.
- 13.7 On a show of hands each person present at the meeting and entitled to vote (whether personally or by Representative) shall have one vote only. On a poll every Holder who is present in person or by Representative shall be entitled to one vote for every nominal \$1.00 of Stock of which he is the holder PROVIDED ALWAYS THAT any Stock for the time being held by Spark NZ or any related company (as defined in Section 2(3) of the Companies Act) of Spark NZ shall not whilst so held confer any right to vote.
- 13.8 In the case of joint Holders the vote of the senior who tenders a vote whether in person or by Representative shall be accepted to the exclusion of the votes of the other joint Holders and for this purpose seniority shall be determined by the order in which the names stand in the Register in respect of the joint holding.

14 SPECIAL RESOLUTION

- 14.1 The expression "Special Resolution" when used in this Deed means a resolution passed at a meeting of Holders duly convened and held in accordance with the provisions herein contained by Holders holding Stock with a Principal of no less than 75% of the Stock Moneys held by those persons who are entitled to vote and who vote on the question and the expression "Special Resolution" referring to any Class of Holders shall have a corresponding meaning in relation to meetings of such Class.
- 14.2 Without limiting the rights powers and discretions conferred on the Supervisor by the Trust Deed and subject to regulation 15 below and in particular to the proviso to regulation 15.1 below, a meeting of Holders shall in addition to all other powers

which by the Trust Deed are specified as exercisable by Special Resolution have the following powers exercisable by Special Resolution namely:

- power to sanction either unconditionally or upon any conditions the release of the Issuer or any other Guaranteeing Group Member from the payment of all or any part of the Stock Moneys;
- (b) power to sanction the exchange of Stock for or the conversion of Stock into shares, stock debentures, debenture stock or other obligations or securities of the Issuer or any other Consolidated Group Member or any other company formed or to be formed;
- power to postpone the day when the Stock shall become payable and to suspend or postpone or, with the concurrence of the Issuer, to accelerate the payment of interest on Stock;
- (d) power to sanction any alteration, release, modification, waiver, variation or compromise or any arrangement in respect of the rights of the Holders against the Issuer and/or any other Guaranteeing Group Member howsoever such rights shall arise;
- (e) power to assent to any alteration or addition to the provisions contained in the Trust Deed or the terms or conditions attaching to the Stock proposed or agreed to by the Issuer and to authorise the Supervisor to concur in and execute any supplemental trust deed embodying any such alteration or addition;
- (f) power to give any sanction, assent, release or waiver of any breach or default by the Issuer and/or any other Guaranteeing Group Member under any of the provisions of the Trust Deed;
- (g) power to discharge, release or exonerate the Supervisor from all liability in respect of any act of commission or omission for which the Supervisor has or may become responsible under the Trust Deed;
- (h) power to sanction any scheme for the reconstruction of the Issuer or any other Guaranteeing Group Member or for the amalgamation of any such person with any other company where such sanction is necessary;
- subject to the provisions of the Trust Deed power to remove any Supervisor and to approve the appointment of or appoint a new Supervisor;
- (j) power to authorise the Supervisor to concur in and execute any supplemental deed or other document embodying such sanction, authority or approval, assent, release, waiver, direction or request.

15 SPECIAL RESOLUTION

15.1 A Special Resolution passed at a meeting of Holders duly convened and held in accordance with the regulations contained in this Fourth Schedule shall be binding

upon all the Holders whether present or not present at the meeting and each of the Holders and the Supervisor (subject to the provisions of its indemnity contained in the Trust Deed) shall be bound to give effect thereto accordingly and the passing of any such resolution shall as between the Supervisor and the Holders be conclusive evidence that the circumstances justify the passing thereof the intention being that it shall rest with the meeting to determine without appeal whether or not the circumstances justify the passing of such resolution PROVIDED ALWAYS THAT a resolution which affects either a particular Holder or a particular Class of Holders only as opposed to the rights of the Holders generally shall not be binding on such Holder or Class of Holders (as the case may be) unless such Holder agrees, or such Class of Holders agree as provided in regulation 17.1 below or otherwise, to be bound by the terms of any such resolution.

- 15.2 A meeting of any Class of Holders shall (without limiting the other powers exercisable by any such meeting) have power by Special Resolution:
 - to sanction any compromise or arrangement proposed to be made between the Issuer and/or the other Guaranteeing Group Members and the relevant Class of Holders affected;
 - (b) to sanction any alteration, abrogation, modification or waiver of the rights of the relevant Class of Holders affected against the Issuer and/or the other Guaranteeing Group Members.

16 MINUTES

Minutes of all resolutions and proceedings at every meeting shall be made by the Supervisor or if the Supervisor shall not be present at any meeting by some person appointed by the chairman of such meeting and duly entered in books from time to time provided for that purpose by the Supervisor at the expense of Spark NZ and any such minutes as aforesaid if purporting to be signed by the chairman of the meeting at which such resolutions were passed or proceeding had or by any persons appointed by the said chairman of the meeting for the purpose or by the chairman of the next succeeding meeting of Holders shall be prima facie evidence of the matters therein stated and until the contrary is proved every such meeting in respect of the proceedings of which minutes have been made shall be deemed to have been duly held and convened and all resolutions passed or proceedings had thereat to be duly passed and had. Copies of such minutes shall be furnished by the Supervisor to Spark NZ as early as possible after the holding of the meeting to which they refer.

17 **CLASS**

A resolution which in the opinion of the Supervisor affects one Class of Holders only (and which does not affect any other Holders) shall be deemed to have been duly passed if passed at a separate meeting of the Holders of Stock of that Class and all the preceding provisions of this Fourth Schedule shall apply to such meetings mutatis mutandis as though references therein to Stock and Holders were references to the Stock of the Class in question and to the Holders of such Stock respectively. *** End of Schedules ***

Signed for and on behalf of Spark New Zealand Limited by:

Director

Director/Secretary

The New Zealand Guardian Trust Company Limited by:

Authorised Signatory

Authorised Signatory